# **Supplier Integration Systems**

# Understanding Adoption Process and Structure

Master's thesis Christian Bercz April 2008







### **Supplier Integration Systems**

Understanding Adoption Process and Structure

April 2008

Master's thesis Industrial Engineering & Management School of Management and Governance University of Twente, Enschede, the Netherlands

By Christian Bercz Student number: 0005843

Supervisors

Prof. dr. Jos van Hillegersberg, Dr. ir. Fredo Schotanus School of Management and Governance, University of Twente

Ir. Paul van Wilgen Capgemini

### Preface

The inspiration for this master's thesis sparked at the course called Business Process Integration Lab. This course was a part of the master track Information Systems & Change Management of the master Industrial Engineering & Management at the University of Twente. The coursed about e-business encompassed all the ingredients that make the master interesting. A combination of business and technology, it covered economics, process management, tactical and strategic issues, and the complexities and opportunities of new technologies in business. When I had to arrange my master's assignment the contents of this course quickly popped up as an interesting topic.

The professor teaching the course, Jos van Hillegersberg, recognized my enthusiasm and provided me with some contacts. Capgemini was quick to respond to my application as my proposal fitted right in their research need. The result is this master's thesis.

As many of my fellow students could agree with, writing your master's thesis is a process of many ups and downs, but in the end a worthwhile learning experience. Without the help of others this would not have been possible. My supervisors at the University of Twente provided me with valuable feedback and guidance. They complemented each other as Fredo took up the role of a critical reviewer and Jos the role of a guide providing valuable directions. Also I would like to thank all the respondents for their cooperation and input for my case studies.

This thesis also marks the end of a great period as a student. Without the moral and financial support of my parents and the "moral" support of my friends, I don't think it would have been the same.

Last but not least Capgemini and especially my supervisor Paul deserve many credits for their support. They provided me with the means to do my work, valuable contacts, a positive working environment, a relevant assignment and valuable feedback. But most of all the sincere interest in me and my work was appreciated. While my studies are at an end, I am fortunate to be able to continue working at Capgemini as a consultant. I look forward to more learning experiences and to put theory into practice.

### Management summary

#### Introduction

The challenge of coordinating all the actors of an increasingly complex supply chain, due to increasing globalization and growth, and the pressure to reduce costs, to improve efficiency, and to enhance customer satisfaction has stimulated manufacturing companies to implement a Supplier Integration System. These are inter-organizational systems that aim at the integration of suppliers. They are enablers of Supply Chain Management and are used in operational purchasing and planning of direct supplies in manufacturing firms.

Supply Chain Management thinking and the rise of e-business have significantly increased the diversity of Supplier Integration System structures. How companies arrive at these solutions can be described in an adoption process model. The impact of the adoption process on the structure is the topic of this study. This study extends an existing adoption model in order to better reflect the adoption of diverse structures.

#### **Research method**

To study the impact of the adoption process on the structure, eleven cases have been examined from the perspective of the purchasing manufacturing firm. The studied manufacturing firms are large global multinationals with at least 5000 employees and annual sales exceeding 2 billion Euros. Input for the cases comes from semi-structured interviews and referred and public documents about the company.

#### **Supplier Integration System structure**

The case studies indicated that Supplier Integration System structures can best be described in terms of three basic structures, complemented by the aspects of the structure; breadth (the % of partners integrated), volume (the % of automated information flow), diversity (the % of shared information and processes), technology (type of software, network and data standards) and context (relationship and supply type). The three observed basic structures have distinct governance structures, different advantages and are used in a complimentary way for both internal and external supply chain integration.

1. In the system-to-system structure, partner systems are directly interfaced. This structure is considered most difficult to implement, but it enabled a high degree of integration.

2. In the VAN structure an electronic integration hub is positioned between the involved parties. This hub can be provided by an external integration service provider. This structure enables a high degree of integration and large networks, but it makes the company dependent on a sometimes costly and not always competent service provider.

3. In the supplier portal structure an integrated website provides partners secured access to the data. It is the most accessible structure, but only provides partial integration, thus still requiring manual tasks.

### **Adoption process**

The case studies could confirm most elements of the adoption process model used in this study. They showed long term projects with significant changes to the structure during the adoption and an unsettled and indecisive nature of the factors that may lead to adoption. The case companies are not in total control of their environment, but are not helpless victims either. Also the inter-firm level appeared to be an important part of the structure where the company could exert its influence.

Additionally the case studies showed diverse and dynamic structures and an evolving, intangible and iterative adoption process. Also it was noticed that an implemented Supplier Integration System structure impacts the organization and its environment and may lead to new innovations in the technologies used. These observations are used to extend the existing model.

#### Adoption process impacts on structure

While no conclusive answer can be given on the complete impact of the adoption process on the structure, the cases indicate some impacts, concerning specific aspects of the structure. The basic structure was mainly determined by the perceived nature of the technology and the perceived relative advantages. The cooperation, to achieve higher breadth, volume and diversity, was stimulated through persuasion tactics in a climate of trust, and coercion tactics using buyer's power. Also the skill and infrastructure of all the involved parties, the existence and use of an industry standard, the execution flexibility and the basic structure type were import conditions for achieving higher breadth, volume and diversity. Industry standards are actively promoted in industry associations. Experience with the system and widely diffused industry standards were reported to enhance flexibility. Concerning the technology path dependency and network effects are confirmed by the cases.

#### Recommendations

The extended adoption model can be used by practitioners to analyze their situation. Based on the model they can identify the conditions, which are not met, and develop deliberate actions (such as coercion and persuasion tactics) to meet them. Changes in the structure should be considered, based on the costs and benefits of these deliberate actions and alternative opportunistic behavior (seizing the right conditions at the right moment).

# Contents

1	Introduction	8
2	Research Context	10
	2.1 Purchasing becomes Supply Management	10
	2.1.1 Purchasing defined	10
	2.1.2 Supply Management and Supply Chain Management	10
	2.1.3 Collaborative buyer-seller relationships	11
	2.2 E-business	12
	2.2.1 Impact of e-business	12
	2.3 Adoption and Structure of Supplier Integration Systems	13
	2.3.1 Structure	13
	2.3.2 Adoption Process	13
	2.4 Conclusion	13
3	Research Design	14
	3.1 Research contribution	14
	3.1.1 Structure	14
	3.1.2 Adoption process	14
	3.1.3 Conclusion: contribution	16
	3.2 Research Questions	16
	3.3 Research Method	17
	3.3.1 Literature study	17
	3.3.2 Case Studies	17
	3.4 Organization of the thesis	19
4	E-procurement and the Supplier Integration System	21
	4.1 E-procurement defined	21
	4.2 Distinguishing e-procurement types	22
	4.2.1 Characteristics that describe e-procurement types	22
	4.2.2 A classification of e-procurement types	23
	4.3 The Supplier Integration System explained	25
	4.3.1 General Supplier Integration System characteristics	25
	4.3.2 What Supplier Integration Systems do: Processes	27
	4.4 Distinguishing Supplier Integration System structures	27
	4.4.1 What: Information and Process perspective	28
	4.4.2 How: Technology perspective	30
	4.4.3 Context: Inter-Organizational perspective	30
	4.5 Conclusion	31
5		32
	5.1 Perspectives on IOS adoption	33
	5.2 The factor approach to IOS adoption	33
	5.3 The processual approach to IOS adoption	35
	5.4 Adoption and the structure of Supplier Integration Systems	36
	5.4.1 Processes and Information sharing	37
	5.4.2 Standards	37
	5.5 Conclusion	38
6		39
	6.1 Basic structures	39
	6.1.1 System-to-system structure	40
	6.1.2 VAN structure	41
	6.1.3 Supplier Portal structure	42
	6.1.4 Integration focus	43
	6.1.5 Governance structures and process architecture	43
	6.1.6 Case observations	44

6.2 Breadth and volume	46
6.2.1 Evolving size	46
6.2.2 Achieving higher breadth and volume	46
6.2.3 Case observations	47
6.3 Process and Information diversity	48
6.3.1 Diversity	48
6.3.2 Processes	48
6.3.3 Information types	50
6.4 System Intelligence and Integration level	51
6.4.1 Case observations	51
6.5 Technology	52
6.5.1 Case observations	53
6.6 Supply and Relationship Context	54
6.6.1 Supply Context	54
6.6.2 Relationship context	54
6.6.3 Case observations	55
6.7 Adoption process	57
6.7.1 Awareness and attitude formation	57
6.7.2 Decision: costs, risks and benefits	57
6.7.3 Implementation	58
6.7.4 Case summary on adoption	59
6.8 Case conclusions	59
7 Conclusion and Recommendations	62
7.1 Conclusions	62
7.1.1 Structure	62
7.1.2 Adoption process	63
7.1.3 Adoption process impacts on structure	66
7.1.4 Limitations	67
7.2 Recommendations	68
7.2.1 Practitioners	68
7.2.2 Future research	69
8 References	71

# List of figures and tables

Figure 1: Screenshot of a Supplier Integration System (www.e2open.com)	8
Figure 2: Simplified TOE framework	15
Figure 3: Simplified adoption process model [24]	15
Figure 4: Simplified adoption process model including contribution	16
Figure 5: Overview of the research design	
Figure 6: Purchasing process model ([3], p15)	24
Figure 7: Positioning of e-procurement types (with adaptations [12])	24
Figure 8: Supplier Integration System	
Figure 9: TOE Factor model of IOS adoption [24]	34
Figure 10: Second order model of IOS adoption (with some adaptations [24])	
Figure 11: System-to-system structure	40
Figure 12: VAN structure	41
Figure 13: Supplier portal structure	42
Figure 14: Adoption Process (with adaptations [13])	64
Figure 15: Extended adoption process model	66
Table 1: Characteristics of buyer/seller relationships [2]	11
Table 2: Case Companies	19
Table 3: Respondents	19
Table 4: Classification Characteristics of e-procurement	23
Table 5: SCOR level 3 source processes (with adaptations) [47]	
Table 6: Collaborative Planning processes [48]	29
Table 7: Information types	29
Table 8: Functional aspects	30
Table 9: Technological and structural aspects	30
Table 10: Inter-organizational characteristics	31
Table 11: General characteristics of the Supplier Integration System	
Table 12: Basic structure and integration focus (part 1)	44
Table 13: Basic structure and integration focus (part 2)	
Table 14: Governance structure	
Table 15: Breadth and Volume	47
Table 16: Diversity	48
Table 17: Processes supported	
Table 18: Information shared	
Table 19: System intelligence & integration level	51
Table 20: Technology	53
Table 21: Supply context	55
Table 22: Relationship context	
Table 23: Benefits	58
Table 24: Costs and risks	58
Table 25: Summarized case observations (part 1)	
Table 26: Summarized case conclusions (part 2)	61

## **1** Introduction

The internet-hype has settled down, but still the potential rewards of doing business electronically are considered huge. Improved inter-firm collaboration and integration with suppliers are among them. An example is the way STMicroelectronics (one of the world's largest semiconductor companies) faces its business challenges with system-to-system integration of contract manufacturers using RosettaNet standards. In a 2005 member's whitepaper of the European B2B forum for the Electronics Industry they share their experiences of a global solution [1]. They report that "the challenge of coordinating all the actors of an increasingly complex supply chain", due to increasing globalization and growth, and "the pressure for semiconductor companies to reduce costs, to improve efficiency, and to enhance customer satisfaction", have stimulated STMicroelectronics to implement their Supplier Integration System. The reported project outputs were "a strong improvement of real-time visibility of the manufacturing process, the automation of shipments and the improvement of data quality and productivity."

This example illustrates that e-business is a part of modern business. STMicroelectronics regards electronic communications and automated business processes with their partners essential to achieve large volumes and to compete with efficient business processes. Many other mass producing manufacturing companies are following the same line of thinking when they consider e-business in the supply chain in general and more specific in purchasing. Also software companies, such as e2open, are specializing in these kinds of systems. Figure 1 shows a screenshot of such a Supplier Integration System.

e	~	ESOBEN.				Soard Velos	n Bayer (Regular)	: boerd_user	Logout	1 Help (
co	Inte	orations: «« Salad Co	Rebaretick View > Col aboretic	Fiter - Col	Inthis received with	isher > Collabor	stan Selector > C	ollabora		
Rapid	a P	fulti-Colleb View Surre	nary							
Nult	I-CU	Balanatius View - Supp	oly View							7 IB G
Sect	Date	e: 10/15/2805	Bid Date	10/15/2016	10	Calendari	Cefoult .	beg	end:	
Auge	har:	Repidity Dectronics					6-11-10-20-00-0-00-00-00-00-00-00-00-00-00-00-			
								17	uentory Report	Update
								11		
A	1	presset Order In	ventory Performance Recto							
Coll	abs :	1-3.083								100
г		Soupler Customer Item Item	Data Neasona 🗇 👻	10/10/05	38/28/05	11/21-05	11/22/05	10/25/35	10/24/05	350
E	13	CH059100 BH-CH359	Porscalt:	120 :	120	140	308	340 -	110	12 *
			Quynt	100	120	110	03	141	110	32.0
			Supply Shortege	-20	8	-30	-320	0	D.	0
			Supply Shorbege Ecception	7.1-2		000	all states of	177-		
			Currulative Forecast	330	249	350	650	6.23	930	4,9
			Cumutetive Commit	100	220	320	-11.0	\$\$3	660	785
		0	Cumulative Supply Shorts de	-20	-20	-50	-270	-270	-278	-27
			Cumulative Shartage Exception							
1	8	CM059100 8M-CM059	Parecest.	70	100	1.55	138	120	110	10
			Commit	70	160	145	110	123	110	300
			supply shortage	U	-20	-10	-20	W	U.	U.
			Supply Shortage Exception	100						100
		0	Cumulative forecast	70	370	325	455	575	663	785
			Cumulative Commit	30	150	255	425	525	633	735
			Cumulative Busply Shortage	0	-20	-30	-5D	-61	-320	- 50 +
4		*		4						

Figure 1: Screenshot of a Supplier Integration System (www.e2open.com)

This has drawn our attention resulting in the topic of this thesis: *Supplier Integration Systems*. These are inter-organizational systems that aim at the integration of suppliers. They are enablers of Supply Chain Management and are used in operational purchasing and planning of direct supplies in manufacturing firms.

The research project is part of the master's program of Industrial Engineering and Management at the University of Twente. The assignment is commissioned and facilitated by Capgemini Consulting Services. As a leading consulting firm, Capgemini needs to be on top of the latest developments in the business world and therefore is constantly executing and facilitating practice oriented research. This master's thesis aims to contribute to Capgemini's research on collaboration and integration in the supply chain. The development of increasing integration in the upstream supply chain reflects a growing emphasis concerning the importance of suppliers. As part of the Consulting practice of Supply Chain Management the cluster Procurement is particularly interested in how its customers today are using technology to enable even tighter collaboration with their suppliers.

Although the idea of integration is not new, the execution of e-Business enabled integration in practice is not completely understood. It is recognized that Supplier Integration Systems are a class of systems that can have various structures. What structures are used in practice and why companies have these structures is not well understood. The process of arriving at a certain Supplier Integration System structure, called the adoption process, can explain why companies have their particular structures. New insights in this area would enable Capgemini to make better service offerings to potential customers.

# Therefore this study aims at understanding the adoption process and its impact on the structure of Supplier Integration Systems in manufacturing companies.

To enable a fruitful study of this topic, we will first explore the research context in Chapter 2. This context forms the basis of the research design in Chapter 3. The research design covers the research contribution, - questions and - method. This is followed by a more elaborate outline of this thesis at the end of Chapter 3. In Chapter 4 the structure of Supplier Integration Systems is explored from a theoretical point of view. In Chapter 5 the theory on adoption processes is discussed. Chapter 6 covers the empirical part of the thesis. The adoption process and its impact on the structure in practice are studied with case study research. In Chapter 7 conclusions will be presented and the results and its implications will be discussed.

# 2 Research Context

The context of this study's topic is shaped by two closely related developments of the last twenty years. The first development is the elevation of the purchasing function and the second is about the rise of e-business. These two developments need to be understood first because they represent the business context and the IT context of Supplier Integration Systems (Sections 2.1 and 2.2.) The structures of Supplier Integration Systems are formed during the adoption processes within this business and IT context. This process and its resulting structure will be further explored in later chapters. An initial explanation of these important concepts is provided here in Section 2.3.

### 2.1 Purchasing becomes Supply Management

The elevation of the purchasing function from a purely transactional mindset towards the recognition of the strategic importance of purchasing and a collaborative approach to suppliers, explain the business context of Supplier Integration Systems. This section explains this development starting with purchasing and its definition. The section ends with supply management and the rise of the collaborative approach to suppliers. An approach that is supported by Supplier Integration Systems and at the same time is a necessary condition to make successful use of them.

### 2.1.1 Purchasing defined

No organization is completely self sufficient and therefore has to rely on external sources for its organizational needs. Purchasing is about the fulfillment of these needs. During the last decades the subject has received increasing attention both in literature and in practice. From this attention different terms, concepts and definitions have evolved. Monczka [2] for instance uses the term purchasing for the functional group in the organizational chart and for the functional activity of buying. In this thesis the definition of purchasing by van Weele [3] (p14) will be used because of its richer detail.

Purchasing is defined as: "obtaining from external sources all goods, services, capabilities and knowledge which are necessary for running, maintaining and managing the company's primary and support activities at the most favorable conditions"

This definition makes clear that purchasing is not only about purchasing materials or office appliances. Both direct and indirect goods and services are included in the definition. Furthermore it involves all activities in the purchasing process including administrative and managerial processes, and it relates to all departments involved in purchasing.

### 2.1.2 Supply Management and Supply Chain Management

According to some a firm has to focus on a few activities tot achieve competitive advantage in the long-term [4]. None core activities should then be outsourced to specialized suppliers. Nowadays even many large parts of primary activities such as production are outsourced to Contract Manufacturers, such as in the example of STMicroelectronics. This in part can explain the high and rising purchasing to sales ratio, which for industrial companies can be up to 70%. [3]. This development of increasing purchasing value and increased outsourcing of essential activities caused a higher awareness of the importance of purchasing to business.

The notion that purchasing, as any other business function, has to excel and contribute to competitive advantage was already stressed in the late 70s and early 80s. Kraljic [5] called for a total change of perspective: from purchasing as an operating routine to supply management

as a strategic function. Monczka [2](p8) follows this perspective as he defines supply management as "the process of identifying, evaluating, selecting, managing, and developing suppliers to realize supply chain performance that is better than that of competitors". It is cross-functional as it involves different business functions in make-or-buy decisions, engineering, quality assurance, operational purchasing and it involves the suppliers themselves. Because its objective is supply chain performance, it is closely related to Supply Chain Management (SCM).

SCM is defined as: "the task of integrating organizational units along a supply chain and coordinating material, information and financial flows in order to fulfill ultimate customer demands with the aim of improving competitiveness of a supply chain as a whole" [6](p11).

Supplier Integration Systems enable upstream SCM as they support the coordination of material, information and financial flows between manufacturing companies and their suppliers.

### 2.1.3 Collaborative buyer-seller relationships

The concepts of collaboration, integration and coordination are closely related. Integration and coordination are regarded as the two pillars of SCM and collaboration as an important building block [6]. Monczka [2] distinguishes the collaborative approach from the traditional approach to suppliers with the characteristics in Table 1. The traditional approach is the case where the marketplace is characterized by strong buyers' power and a purely transactional approach to the relationship.

	Traditional Approach	Collaborative Approach
Suppliers	Multiple sources played off against each other	One or a few preferred supplier for each major item
Cost Sharing	Buyer takes all cost savings when possible; Supplier hides cost savings	Win-win shared rewards
Joint Improvement Efforts	Little or none	Joint improvement driven by mutual interdependence
Dispute resolution	Buyer unilaterally resolves disputes	Existence of conflict-resolution mechanisms
Communication	Minimal or no two-way exchange of information	Open and complete exchange of information
Marketplace Adjustments	Buyer determines response to changing conditions	Buyer and Seller work together to adapt to changing conditions
Quality	Buyer inspects at receipt	Designed into the product

#### Table 1: Characteristics of buyer/seller relationships [2]

Collaboration between buyers and sellers today takes place in engineering, planning, marketing and logistics. Although the collaborative approach is increasingly applied, which is illustrated by the trend in business towards fewer suppliers and a higher degree of buyer-seller collaboration [7], this does not mean that it is always the best approach. It has both advantages and disadvantages. Advantages are a higher level of trust and long-term contracts, both leading to joint efforts resulting in a more efficient and agile value chain [2]. But disadvantages are that partnerships are expensive to develop and maintain, and that they cause the risk of supplier lock-in [3]. The choice of partners and the choice to which extend and how to collaborate with them are an important part of supply management.

### 2.2 E-business

The rise of e-business forms the IT context of Supplier Integration Systems, since Supplier Integration Systems are examples of e-business solutions.

E-business can be defined as: "the conduct of automated business transactions by means of electronic communications networks end-to-end" [8].

It means to automate and integrate business processes across all the members of an extended supply chain. Therefore it is about collaboration and coordination with business partners, including suppliers. The crucial difference with information systems in general is that ebusiness is about supporting intra-organizational and inter-organizational relationships. Ebusiness therefore crosses organizational boundaries while most information systems support business processes and/or functions internally. The integration of separate information systems and business processes internally has long been an important development. Enterprise resource planning (ERP) systems, which are large, integrated business transaction processing and reporting systems, are a well known example. E-business aims to integrate business processes and information systems of different companies in a supply chain [8]. This fits in the tradition of inter-organizational systems (IOS).

IOS are defined as: "planned and managed ventures to develop and use IT-based and/or human-based information exchange systems to support collaboration and strategic alliances between otherwise independent actors" [9].

E-business solutions are also called e-business systems or networked systems.

### 2.2.1 Impact of e-business

The automation and integration of business processes across all the members of an extended supply chain has several advantages. Improved operational efficiency and reduction in operating costs are just some of the recognized benefits [8]. Another major recognized benefit is the reduction of the bullwhip effect [10]. In general e-business is thought to reduce transaction costs and at the same time achieve a higher level of coordination in markets, normally associated with hierarchies. It is therefore predicted to increase the degree of outsourcing. Also long-term relationships with fewer suppliers are stimulated due to higher investment costs in the relationship and progressive benefits from a longer learning curve [8].

But inter-organizational systems, which support information sharing and automate processes, are not new. Already in the early days of information technology, the 60s, the idea of information systems crossing organizational boundaries was pointed out [11]. Also the internet finds its origins in these days [12]. Inter-organizational systems therefore are not new. In the late 1980s Electronic Data Interchange (EDI) emerged to connect the new Manufacturing Resource Planning (MRP) and ERP systems to supplier's systems in order to deliver orders and schedules for direct items.

With the breakthrough of the internet in the 1990s, e-business came into use. Because the underlying technologies have developed significantly, the impact of e-business can be found in the increased possibilities, enabling more effective Supply Chain Management. Modern e-business technologies have some crucial advantages that make a difference when integration in the supply chain is pursued. Increased reach, range, speed and agility would summarize these crucial advantages [8]. Reach refers to number of actors and their locations that can be connected. Range is about the functionalities in terms of the processes the system supports and the information that is shared. Speed is about the timing (e.g. real-time or batch sharing of information) and level of integration with back end systems. Agility refers to the ease of

making changes to the system in terms of functionality or in terms of scale (e.g. adding a new partner). These advantages have triggered the adoption of new e-business solutions and migration to new technologies, enabling more effective Supply Chain Management. As a result the diversity of Supplier Integration System structures has increased significantly.

### 2.3 Adoption and Structure of Supplier Integration Systems

This study aims at understanding the adoption process and its impact on the structure of Supplier Integration Systems. In later chapters this will be addressed in more detail but in this section an initial explanation of the concepts structure and adoption process is provided.

### 2.3.1 Structure

The structure refers to the way the Supplier Integration System is composed of its individual components. These components are the business and technological aspects of the system. It is about how the information system looks like. This includes aspects such as: the system architecture, the information technologies, the information types, the business processes, the users of the system and its governance structure. The structure is initially a result of its adoption process including all the decisions made during this process. Later during its "product life cycle" decisions due to evaluations or changes in the system environment may lead to changes to its structure.

### 2.3.2 Adoption Process

Adoption is a demand side concept that entails the acceptance of an innovation (e.g. the Supplier Integration System). Innovation "refers both to the output and the process of arriving at a technological feasible solution to a problem triggered by a technological opportunity or customer need" (p68 [13]). This assumes that the person or organization is capable of adoption of the innovation. The adoption process starts when the organization becomes aware of the innovation. Based on this attitude formation a decision is made about the adoption. During the implementation the innovation is put into use. Finally in the confirmation phase the decision is evaluated. This adoption process is not a straightforward process. On the contrary it is a very diverse, complex and dynamic process involving numerous actors and iterations [13]. It is often a problem solving or opportunity driven long term endeavor and it is influenced by many different factors.

### 2.4 Conclusion

Supply Chain Management thinking and the rise of e-business have significantly increased the diversity of Supplier Integration System structures. New structures have been adopted, while at the same time older technologies, such as traditional EDI are not disregarded [8]. In this chapter it was pointed out that companies are increasingly outsourcing production, such as in the STMicroelectronics example. They are reducing their supplier base and are investing in increasingly long term relationships. Such developments stimulated companies to adopt varying e-business technologies to integrate their processes with their suppliers. The adoption processes of these inter-organizational systems are diverse, complex and dynamic. The outcomes are not always the same, as projects fail or are disregarded, and when they lead to an implementation the structure is very diverse.

### 3 Research Design

This section explains the set-up of this research project. In Chapter 2 it was concluded that Supply Chain Management thinking and the rise of e-business have significantly increased the diversity of Supplier Integration System structures. These structures are, at first, the result of their adoption process. The study aims at understanding the adoption process and its impact on the structure of Supplier Integration Systems. In Section 3.1 the research contribution to practitioners and particularly to theory will be explained. From this paragraph the central research question is derived. In order to answer the central research questions are explained. To answer these questions, knowledge, derived from literature and practice, is required. Section 3.3 will elaborate on the actions that have been performed to acquire this knowledge. This chapter will finish with a more elaborate outline of the rest of this thesis.

### 3.1 Research contribution

As the STMicroelectronics example illustrates, the business challenges and opportunities requiring integration and the possible benefits of Supplier Integration Systems, justify the efforts. Although there is much useful literature, theory falls short in some areas of Supplier Integration system structures and adoption processes. In this section these gaps in the literature are detected. These gaps concern both structure and adoption process.

### 3.1.1 Structure

Supplier Integration Systems are a sub-class of IOS that support vertical integration in the operational purchasing and planning process for manufacturing firms' direct supply. Implementation of Supplier Integration Systems, as an example of e-business, have endured the internet-hype and turbulent and rapid developments [8]. This has led to two problems.

The first problem is that these Supplier Integration Systems are ill-defined and have different names, which have different meanings to different people. Names used for instance are Web-ERP [12, 14], B2B electronic marketplace [15-17], Supplier Portal [18], Collaborative portals [19], e-Hub [15, 16, 20, 21], extranets [22] or Supply Hub [23].

The second problem is that these systems have very diverse structures, while many theories about these systems ignore these distinctions (e.g. [24-26]). While the first problem is just an inconvenient semantic problem, this generalization is an oversimplification of reality that cannot be ignored as different conditions and objectives require different Supplier Integration System structures.

Because of this a systematic description of the variety of structures is required. There are many classifications (see Chapter 4) but they all describe different aspects of this structure and the resulting classifications have a relative high level of abstraction. The diversity of networked systems is addressed in the inaugural lecture of van Hillegersberg [27], which resulted in a more complete list of characteristics that can be used to describe the diversity of structures. This line of thinking is therefore followed in this thesis, by developing a way to describe and distinguish Supplier Integration Systems structures in more detail.

### 3.1.2 Adoption process

The adoption of networked systems has been studied extensively before [24, 28-30], leading to factor and process models of adoption. The factor models typically categorize factors, which influence organizational actions leading to adopting or reject the system, in a technology–organization–environment (TOE) framework [24, 26]. This has resulted in a considerable list of factors that includes enablers and inhibitors of adoption and includes the

implicit and explicit motivations of companies to adopt or reject a system. For instance the lack of an adequate IT capability is an inhibitor in the organization category, while in the technology category the perceived efficiency gains of the new technology could be a strong explicit motivation for the company to adopt the system. In Figure 2 a simplified TOE framework is represented. This model will be discussed in more detail in Chapter 5.



Figure 2: Simplified TOE framework

Process models extend this TOE framework with a more comprehensive view as they assume that organizational actions also impact these TOE factors, that there must be a distinction between environmental factors and industry factors, and that adoption processes are complex dynamic longitudinal processes [24]. In Figure 3 the process model is depicted in a simplified representation. This model will be discussed in more detail in Chapter 5.



Figure 3: Simplified adoption process model [24]

But most studies still assume a dichotomous outcome. Either the organization adopts the system or it does not. This is a simplification of reality that was useful for studying the rise of e-business. It helped managers to see potential benefits and barriers and to decide whether or not to go along with the new technologies. But with the ongoing proliferation of Supplier Integration Systems the question shifts from the adoption decision to the decision what type and structure is most suited.

Some studies have studied the impact of the adoption process on some aspects of the structure. Already in studies of traditional EDI adoption, the use of these systems was taken into account [30, 31]. The use of EDI was measured in terms of volume (% of total messages send with EDI), breadth (% of total partners connected with EDI) and diversity (% of total document types supported by EDI). These measures provide a good picture of the range and reach of the networked system. Others have focused on the technological aspects [32]. Here this line of thinking is extended by investigating the structure in more detail as was argued for in Section 3.1.1. Then the relation between the adoption process and the structure becomes interesting again. The more comprehensive process model of adoption [24] is therefore enhanced to accommodate for the impact of the adoption process that is resulting in various structures.

To take into account the diversity of structures the adoption model needs to be enhanced with a more diverse outcome. In Figure 4 a simplified representation of the suggested adoption process model of Supplier Integration Systems is depicted. It builds on the adoption process model of Kurnia and Johnston, which in turn is an extension of the TOE-framework [24] (see Chapter 5). In green various factor categories, that impact organizational actions with regard to adoption, are shown. It can be seen that only the environment category can not (or at least almost not) be influenced by the organization. The other categories can be influenced. In grey the adoption of a specific Supplier Integration System structure is depicted. Taking into account the diversity of these structures is the most important contribution of this study. In Chapter 7 this model will be explained in more detail.



Figure 4: Simplified adoption process model including contribution

### 3.1.3 Conclusion: contribution

The contribution of this study mainly lies in a more detailed view of structure, the adoption process and the impact of the adoption process on the structure. This study will explore a systematic description of Supplier Integration System structures (see Chapter 4) and enhances the process model of adoption.

For practitioners a better understanding of the various structures and the adoption processes would be useful to improve decision making and management of such complex adoption processes. By systematically describing structures this study shows practitioners what options they have when they want to integrate with their suppliers. This study will improve existing adoption models and will show how to use them as a guideline to determine the right conditions to adopt a specific Supplier Integration System structure and what can be done to improve these conditions.

### 3.2 Research Questions

The objective of understanding the impact of adoption process on Supplier Integration System structures can be translated into the following central research question:

# What is the impact of Supplier Integration System adoption processes on Supplier Integration System structures of manufacturing companies?

This main research question is divisible into three parts: 1. Structure, 2. Process, 3. Relation. For each part sub questions have to be answered in order to answer the central research question. The rest of this thesis will answer these questions.

### 1. Structure: What are the various Supplier Integration System structures?

- a. How can Supplier Integration Systems be defined?
- *b.* How can Supplier Integration Systems be distinguished from other e-procurement types?
- c. What are suitable characteristics to distinguish different Supplier Integration System structures?
- d. What Supplier Integration System structures can be observed in practice?

# 2. Process: How can the Supplier Integration System adoption process (best) be studied?

- *a.* What are the different perspectives on adoption processes of inter-organizational systems?
- b. What are the contributions and shortcomings of these approaches?

# **3.** Relation: What is the relation between the adoption process and Supplier Integration System structures?

- a. What theories explain the relation between the adoption process and Supplier Integration System structures?
- b. What adoption processes can be observed in practice?
- c. What explanations of the specific adoption process and its resulting Supplier Integration System structure can be observed in practice?

### 3.3 Research Method

The central research question and its sub questions are answered with findings from a literature study and from an empirical study. The empirical study is based on case study research.

### 3.3.1 Literature study

The literature study is conducted using several online indexes for their advanced search functions and to apply the "snowball method". The snowball method involves using the browsing functions to quickly browse references and citations to find more relevant articles. The online indexes used for the literature study cover at least the top 25 IS journals [33]. Due to the semantic problem concerning Supplier Integration Systems (see Section 3.1.1), many different search terms had to be used resulting in a broad set of literature. The snowball method<sup>1</sup> was used to enhance the results and to find overlooked articles.

### 3.3.2 Case Studies

The empirical study is based on case study research because of two reasons. The first is of a practical nature. The efforts and costs to get a large data set required for quantitative research did not fit within the time and budget constraints of this research project. But more

<sup>&</sup>lt;sup>1</sup> Searching relevant literature by following links to references and to articles citing the article.

importantly the second reason to do qualitative research is because of intrinsic reasons. The adoption process is a complex and dynamic process and the subject of Supplier Integration Systems is not without ambiguity. This means that qualitative research is better suited for this subject. In interviews the meaning of certain questions and answers can be interpreted better. Also the interviews contribute by allowing more richness and detail. The anecdotal data that can be provided by interviews is essential to understand the impact of the adoption process. This argument is supported by the call for a processual view of adoption (see Chapter 5 [24]).

### Method

The case studies are based on face-to-face interviews with 1 or 2 subject matter experts of the case companies and additional information provided by the case company. The interviews can be characterized as semi-structured, unconstraint by time and are reviewed:

- The interviews are semi-structured. This means that the interview is guided by the researcher along the different topics, but that the respondents are allowed the liberty and time to answer questions with their own story and anecdotes. The researcher adepts its questions to the particular situation and he tries to anticipate and respond to the answers of the respondent.
- Interviews were finished when all the topics were sufficiently covered as perceived by both the researcher and the respondent. Time constraints did not cause insufficient coverage of specific topics. On average the interviews lasted about 90 minutes.
- All the interviews are integrally recorded. From these records comparably structured interview reports were made. The interviews were all held in Dutch. To reduce translation errors the interview reports are in English. These interview reports were reviewed by the respondents. The checked for errors, wrong interpretations and missing information. In some cases they consulted colleagues to improve the quality of these reports. The reviewed reports are used as inputs for this thesis.

### Selection of cases

Cases were selected based on the following criteria. These criteria narrowed the selection down to a relative small subset from which an arbitrary choice was made. Time constrained the number of cases to 11. All of the case companies were customers of Capgemini in the past.

- 1. The company must be a *mass producing manufacturing company*. This study focuses on Supplier Integration Systems of manufacturing companies. Issues like contract manufacturing and ensuring on-time supplies for mass production, challenge businesses.
- 2. The company must be *large* (at least 5000 employees). Especially in large companies these types of systems are established. For instance traditional EDI was mainly adopted by larger companies because it was costly and required considerable organizational capabilities [8]. It is expected that also for newer systems larger companies have a leading role.
- 3. The company must be *internationally* operating. Supplier Integration Systems in large manufacturing companies are often applied in an international context. This is a complicating factor and should therefore be the same in all cases.
- 4. The company must be *active in the Netherlands*. For practical reasons the selected case companies have to be active in the Netherlands to allow easier access to potential respondents
- 5. The companies must be from *diverse industries*. In order to pick out industry and supply chain effects companies from a diverse set of industries are selected.

Company	Annual Sales in millions €	# Employees	Strongest Geographic Presence (all are globally active)
Pharma1	5,917	13,900	Europe & East-Asia
Pharma2	2,611	13,700	Europe
Packaging1	7,000	40,000	Europe
Packaging2	11,000	51,000	Europe
Truck1	2,500	9,000	Europe & North America
Truck2	7,500	32,000	Europe
Electronic1	26,976	121,000	Global
Electronic2	72,448	398,000	Global
Electronic3	3,500	24,000	Europe & North America
Electronic4	6,365	37,000	Europe and East-Asia
FMCG	8,100	50,000	North America & Europe

Table 2 lists the basic characteristics of the selected case companies.

**Table 2: Case Companies** 

### Respondents

The respondents are subject matter experts within their company concerning the subject of this study. They are located and approached in two ways. Because of established relations with Capgemini some respondents could be located and approached through Capgemini's contacts. In other cases the respondents were located and approached by phone after consulting the case company's website. Before interviews were planned the objective of the study and some example questions were communicated to find the appropriate respondent.

All respondents are active in logistics, supply chain management, purchasing and/or Information Technology. Most respondents were working in a centralized department or at the plant level. Table 3 lists the job titles of the respondents

Company	Respondent 1	Respondent 2
Pharma1	Senior manager Supply Chain Europe	Senior manager plant logistics
Pharma2	Application manager Supply Chain Management	-
Packaging1	Supply Chain support manager	-
Packaging2	Program Manager	-
Truck1	EDI Manager	EDI coordinator
Truck2	Manager Logistics	Manager Material Planning Truck Assembly
Electronic1	Program manager corporate supply management	-
Electronic2	SCM Program Manager; the Netherlands	-
Electronic3	Director Global Applications	-
Electronic4	Program Manager Purchasing Office	Project Manager Purchasing Office
FMCG	Business Process Manager IT services	-

**Table 3: Respondents** 

### 3.4 Organization of the thesis

It was explained that there are three important aspects of the central research question: Structure, Adoption Process, and the relation between the two. These aspects are studied from a theoretical and empirical point of view. This thesis is therefore organized according to these aspects and views. Based on literature, Chapter 4 will cover the structure of Supplier Integration System from a theoretical point of view. In the same way the adoption process and its relation to the structure is covered in Chapter 5. These two theoretical chapters are input for the case studies. They are used as a guideline for the interviews. The empirical part of the study is covered in Chapter 6. The case analysis treats the structure, adoption process and the relation between two in practice. Chapter 7 will synthesize theory and practice into conclusions. This chapter will also provide some discussion of the study and its results. In Figure 5 this outline is summarized. Figure 5 schematically depicts the overview of the research design.



Figure 5: Overview of the research design

## 4 E-procurement and the Supplier Integration System

Without support of Information and Communication Technology (ICT) purchasing as it happens today cannot exist. As in the rest of the organization the purchasing function is supported by many types of applications and internet technology. This chapter presents an overview of the applications that support purchasing and positions the Supplier Integration System among these systems. This chapter answers the first sub-question, concerning the structure, from a theoretical point of view. In Chapter 6 this will be complemented by the observations in practice. The combination will then explain the various structures. Here the sub questions 1a, 1b, and 1c, are treated:

### 1. Structure: What are the various Supplier Integration System Structures?

- a. How can Supplier Integration Systems be defined?
- b. How can Supplier Integration Systems be distinguished from other e-procurement types?
- c. What are suitable characteristics to distinguish different Supplier Integration System structures?

In the world of applications as much views and definitions exist as in management. In this chapter choices are made for certain views and definitions to delineate this research project. E-procurement is defined first in Section 4.1, followed by a classification of e-procurement types in Section 4.2 and finally an explanation of Supplier Integration Systems in terms of the e-procurement classifications in Section 4.3 These three sections combined answer the subquestions 1a and 1b. In this way it will become clear what an Integration System is and how to distinguish it. In Section 4.4, answering question 1c, characteristics are identified to distinguish different Supplier Integration System structures. These characteristics are used in the case studies to guide the interview and to obtain a comparable description of the system structures.

### 4.1 E-procurement defined

E-procurement is a particular form of e-business aimed at supporting purchasing. In Chapter 2 both purchasing and e-business were explained as two related subjects in development. They are the basis for understanding e-procurement.

With purchasing and e-business in mind e-procurement can be defined in a narrow and a broad sense.

- 1. "In a narrow sense it entails information technology solutions for ordering, logistics and handling systems, as well as for payment systems." [3](p175).
- 2. In a broad sense it entails "the use of web-technology in the purchasing process" [3, 12]. Davila et al. [34] also distinguish these two definitions by labeling it in a narrow sense as "e-procurement software" and in a broad sense as "e-procurement technology".

The narrow definition is based on business functions related to purchasing, while the broad definition is explicitly based on a particular technology (web-technology) and the purchasing process. The narrow definition also includes systems used by the purchasing function that do not cross organizational boundaries or do not even interface with external systems. The broad definition explicitly mentions "the use of web-technology" and excludes older electronic

communication network technologies, such as Electronic Data Interchange (EDI) using a Value Added Networks (VAN) or dedicated communication channels. In practice though there are still many implementations using these older technologies. In this thesis we will use the broad definition of e-procurement, but without excluding older collaborative technologies. In this way e-procurement fits within the definition of e-business, since that definition speaks of electronic communication networks. But we want to exclude fax and phone from our definition. Therefore e-procurement is defined in this study as:

### The use of inter-organizational computer networks in the purchasing process.

This definition includes computer networks that cross organizational boundaries, such as the internet and EDI, and excludes networks used only internally, such as intranets, no matter what computer network technology is used. In this way the definition is more robust with regards to the development of new network technologies and fits within the definition of e-business.

Although e-business can be explained in simple terms as "a technology that promotes interenterprise business relations"[8], e-procurement is not to be confused with Supplier Relationship Management (SRM). SRM has been confusingly defined by software vendors and in literature as some integrating suite of e-procurement software that could be viewed as a counterpart of customer relationship management (CRM), while SRM should be viewed as the management of the right relationship with the right supplier [35].

### 4.2 Distinguishing e-procurement types

Now that e-procurement is defined, this section will explain how to distinguish e-procurement types as it encompasses a class of information systems that includes Supplier Integration Systems. This makes clear what systems are related to Supplier Integration Systems and it helps to explain what they are and what they are not. The characteristics identified in literature are listed in Section 4.2.1. Some of these characteristics are later used to describe a classification of e-procurement types in Section 4.2.2, some are used to describe Supplier Integration Systems in Section 4.3, and some are used to distinguish different types of them in Section 4.4

### 4.2.1 Characteristics that describe e-procurement types

In literature many characteristics are used to distinguish among e-procurement types. These characteristics describe different aspects of the system. Some authors use the technological aspects, describing how it works, as a measure to classify different systems. Others describe what the system is used for in terms of the business processes that are supported, and others look at the inter-organizational aspect, describing the context of the system. Many authors use combinations of characteristics that describe different aspects. In a structured literature study of academic literature about e-business and e-procurement, these characteristics were identified. They are used explicitly in the identified sources to classify and distinguish e-procurement types. Here they are synthesized into one list in Table 4. In some cases characteristics found are very similar and therefore are grouped under one label. In the first column the characteristic is labeled, in the second column it is described and in the third the sources in literature that use the characteristics are listed.

Classification	Description	Source
characteristic		
Volume	% of document flow	[31]
Diversity	% of linked trading partners	[31]
Breadth	% of document types	[31]
Process / Phase	Purchasing processes or phases that are supported by different e-Procurement	[12, 36-
	types	38]
Product type	Purchasing different types of items have different supporting systems:	[12, 15,
	1 Manufacturing (direct) vs. operating (indirect) inputs	17, 38]
	2 Importance of purchased item	
Collaboration type	Horizontal collaboration (buying / selling groups ) vs. Vertical supply chain	[17, 37]
	collaboration (between buyer and suppliers)	
Network structure	The companies in the supply chain form a network of users of the e-	[16, 39]
	procurement tool. This dimension distinguishes one-to-many, meaning one	
	supplier vs. many buyers, many-to-one, meaning many suppliers vs. one	
	buyer or many-to-many relationships.	
Power distribution	The power in the relationship and who controls and owns the IOS. The IOS	[16, 17,
/Ownership	can be buyer-centric, seller-centric or neutral	36, 38-
/Control		40]
Order type /	Spot vs. systematic orders / Short vs. Long term contracts	[15, 17]
Contract horizon		
Pricing mechanism	How the prices of goods or services are established. Fixed pricing, auctioning,	[15, 41]
	bargaining, brokerage.	
Relationship type	This describes the type and the extend of the relationship. E.g. Arms' length	[38, 40]
	vs. exclusive partnership	
Interdependency	Concerning the use of the IOS: The degree to which users require each other	[42]
	to achieve their goals. Pooled, sequential and reciprocal	
Intermediation	Presence (and role) of a third intermediate party	[36, 39]
Integration /	Level of integration with back-end systems: measured in number of connected	[27, 38,
Timing and delay	back-end systems, amount of shared information and in the timing and delay	40]
	of information exchange (batch – real time)	
Process & Data	The process & data standard used for the messages and joint processes. Open	[27, 32]
standard	vs. closed or less open standards. These standards can be Proprietary -,	
	Industry - or Universal standards. They can be based on older EDI standards	
	or based on newer XML standards.	
Technology	The type of software used for the system. This can be firm-specific,	[27, 38,
specificity /	customized standard or completely standard software. This can also refer to	40, 41]
Execution and	hard coded vs. Web services.	
development agility		1

 Table 4: Classification Characteristics of e-procurement

### 4.2.2 A classification of e-procurement types

A convenient way to start distinguishing e-procurement types is by using the purchasing processes these systems support as a primary characteristic and the product type as a secondary characteristic to classify them[12]. In this way a complete coverage of all eprocurement types is possible. The purchasing processes cover a wide range of activities. The definition of purchasing by van Weele [3] used in this thesis specifically covers the activities depicted in Figure 6. This process model is used by Harink to classify e-procurement types (see Figure 7). Many similar models are described in literature. For instance business action theory distinguishes six phases; the business prerequisite phase, the exposure and contact search phase, the contact establishment and proposal phase, the contractual phase, the fulfillment phase, and the completion phase [43]. But the van Weele model is appealing because of its simplicity and its applicability to industrial buying of direct goods. The activities in this specific model are distinguished as either Tactical or Operational. In the practice of manufacturing firms, for direct goods, usually a contract is created first as the final output of the tactical process and then this contract is called of many times in the operational process for individual orders. It follows inherently from this process model that the various activities are closely connected. The quality of the output of preceding activities highly impacts following activities. But in practice relatively seldom all steps of the purchasing

process are passed through. This depends on the purchasing situation as a new-task situation requires tactical purchasing activities while for re-buy situations a standard contract can be used many times and for modified re-buy situations only some tactical purchasing activities are required [44].



Figure 6: Purchasing process model ([3], p15)

Concerning e-procurement the management and execution of processes are distinguished in the purchasing process model. Each has their own class of e-procurement systems, which are e-informing and e-transacting. E-informing supports the management of the purchasing process and e-transacting supports the execution of the purchasing process. E-informing consists of purchasing intelligence and e-contract management. E-transacting consists of esourcing, Collaborative Product Commerce (CPC), e-tendering, e-reverse auctioning, eordering and finally Web-enabled ERP, which can be considered another name for Supplier Integration Systems. These e-procurement types are positioned in the purchasing process model, which is depicted in Figure 7. In this model the execution (in green) and management processes (in blue) are distinguished. The secondary characteristic, product type, is then used to distinguish between systems that support purchasing of direct items and those that support purchasing of indirect items.



Figure 7: Positioning of e-procurement types (with adaptations [12])

- Purchasing intelligence consists of the uploading of data in order to process it into valuable purchasing information. With this information decision making is supported and the realization of targets is measured on the basis of performance indicators.
- E-contract management is about making information about contracts available to employees. It can also be used for operational management of contracts and follow up.
- E-sourcing concerns identifying new suppliers, products and services, by using webtechnology. The term e-sourcing is confusing, because in practice it is used often to denote a concept that includes all tactical purchasing systems.

- CPC (Collaborative Product Commerce) relates to collaboration in new product design. Different organizations work together on a new product, which is to be produced by the supplier. The design process, involves a lot of communication and information sharing which is supported with web-technology.
- E-tendering is about tendering electronically. It involves developing and sending RFQs / RFPs and assessing the received information.
- E-reverse auctioning can be considered the opposite of an electronic auction. A buying organization makes a need for a product or service known to a number of capable suppliers, for which these suppliers can make an offer during an electronic auction. Competition is mainly based on the lowest price offered.
- E-ordering is about requesting, ordering and fulfillment of indirect goods. Mainly through electronic catalogues.
- Web-enabled ERP or Web-based ERP (Supplier Integration Systems) aims at the operational purchasing process of direct goods. It is about connecting internal ERPs to suppliers and enables Supply Chain Management.

It is important to note that these e-procurement types can be offered in packages, meaning that more then one type of e-procurement can be meant when a single name is used, such as SRM. Besides e-procurement types, Harink also mentions e-marketplaces. In his view these are electronic meeting places where business partners can exchange information (e-informing) and do business (e-transacting). The e-marketplace is in this view a platform to make e-procurement types available. Within this definition of e-marketplace different types can be distinguished [15, 17].

### 4.3 The Supplier Integration System explained

The objective of this section is to present an overview and a general description of Supplier Integration Systems and their characteristics. In Section 4.2 characteristics to distinguish among e-procurement types are listed and based on processes and product type a classification of e-procurement types is presented. Supplier Integration Systems are interorganizational systems (IOS) that support vertical collaboration in the operational purchasing and planning process for manufacturing firms' direct supplies. In terms of Harink's classification they are called web-enabled ERP. This classification as a starting point helps to distinguish the Supplier Integration System from other e-procurement types (recall Figure 7). In Section 4.3.1 general features of Supplier Integration Systems will be explained and visualized. This section will use some characteristics from Table 4 to describe general characteristics. Since the description of web-enabled ERP in the last section was not very detailed, what Supplier Integration Systems do, will be elaborated in Section 4.3.2.

### 4.3.1 General Supplier Integration System characteristics

When recalling Table 4 the first characteristic to distinguish Supplier Integration Systems is the process characteristic. Using this aspect they were distinguished from other e-procurement types as web-enabled ERPs. In Section 4.3.2 this perspective will be elaborated. General characteristics discussed here are item type and collaboration type and network type. Also a representation will be discussed.

*Item type:* Based on the item type distinctively different systems can be observed based. Take for instance electronic catalogues. These online ordering systems, where users can order what maintenance, repair and offices suppliers from a pre-selected catalogue, are essentially different from Supplier Integration Systems, since they are aimed at ordering indirect items while Supplier Integration Systems are used for direct items. That makes them different, because in general direct items are ordered at larger volumes at higher frequencies, are often more specific and more important to the core business [3]. This means that generally for

direct items more centralized control is required, that it is important for logistics, operations and planning, that integration with ERP systems is more important and that impacts (including risks) are stronger.

*Collaboration type and network type* are also two useful characteristics. The Supplier Integration System crosses organizational boundaries in a collaborative buyer-seller relationship. This is called vertical collaboration and is distinctive from horizontal collaboration. Vertical collaboration is about collaboration between succeeding organizations in the supply chain, such as between buyers and sellers. There is a sequential interdependency [42]. Horizontal collaboration is about collaboration between organizations that are on the same level in the supply chain such as in buying groups or purchasing consortia [37]. The network type is a one-to-many relationship. It is about one buyer and a varying number of suppliers. This contrasts with marketplaces where there is a many-to-many network. For the Supplier Integration System it is assumed that the power in the relationship is located at the buyer, since they are the logical initiators of these kinds of systems. This is inherent to the network type as the suppliers of an organization usually do not have a relationship to each other.

Because of these general characteristics, Supplier Integration Systems are defined as *inter*organizational systems (IOS) that support vertical collaboration in the operational purchasing and planning process for manufacturing firms' direct supplies. They are IOS that provide internal and external users customized access to integrated information, applications, business processes and people, in a collaborative buyer-seller relationship. They are enablers of Supply Chain Management (SCM) as the Supplier Integration Systems purpose of information integration and coordination corresponds to the objective of SCM [45].

The Supplier Integration System can be visualized with Figure 8. In this figure one can see that the buyer interfaces with the system and connects with information flows to the suppliers. To allow integration and automated processing by back-end systems the system transforms the data. For this transformation data standards are used. In this way the Supplier Integration System makes sure that it is compatible to interface with the suppliers. Process management is about the application of workflow system technologies to automate business processes. All Supplier Integration Systems have some degree of connectivity, usually some degree of transformation and in its most advanced form some degree of process management.



**Figure 8: Supplier Integration System** 

### 4.3.2 What Supplier Integration Systems do: Processes

The Supplier Integration System supports operational purchasing processes of direct goods and therefore fits in the definition of Web-enabled ERP (see Figure 7). This name suggests that just like ERP, the integration of information systems across business functions is important. Therefore it involves much more then purely purchasing. This means for instance that the Supplier Integration System interfaces with the ERP and interacts with most business functions in the firm. The connection with ERP means that the system can be applied to two types of processes. These are collaborative transaction management and collaborative planning.

*Transaction management* is about the execution and management of operational purchasing. Ordering, expediting and follow-up are basic execution processes. This involves invoicing, fulfillment, verification and quality control. Also the management of returns falls in this category. These activities are repeated very often. Most frequently the purchasing situation is a re-buy situation, where the item is re-ordered based on an existing contract, plans and forecasts.

*Collaborative planning* and control are the supported management processes. This means that management information on operational purchasing is gathered, stored, processed and exchanged. The Supplier Integration System enables real time sharing of information on planning, production and schedules. Also the sharing of demand forecasts, inventory positions, point-of-sale (POS) data and lead times with partners is an important feature. This is called collaborative planning and forecasting. Some collaborative initiatives have been labeled. They all have the same philosophy and only differ in the information that is shared and in the case of Vendor-managed Inventory (VMI) the management principle that is used for replenishment. The basic philosophy prescribes the exchange of information between business partners in order to optimize the supply chain. It is argued that these collaborative planning initiatives are essential to achieve an efficient supply chain [46]. The earliest initiatives, which originated in the retail and grocery manufacturing industry, where called Efficient Consumer Response (ECR), Vendor-managed Inventory (VMI) and continuous replenishment (CR) [46]. ECR is about sharing information in a demand driven supply chain with the objectives of efficient replenishment. VMI is a technique aimed at the same objective, whereby the supplier has the sole responsibility for managing the customer's inventory policy and replenishment process. CR is a business practice where POS data is used to generate sales forecasts. One step further from these first initiatives, in terms of information sharing, is Collaborative Planning, Forecasting and Replenishment (CPFR).

### 4.4 Distinguishing Supplier Integration System structures

The objective of this section is to be able to distinguish Supplier Integration Systems in a meaningful way. This means that characteristics must be determined, based on which to distinguish the implementations. The general characteristics in Section 4.3 only make up a general description and are used to screen for suitable respondents and to communicate and agree on the subject of the interviews. The combinations of the characteristics identified in this section should allow refining of these general descriptions into more detailed subclassifications. This section will therefore deliver a list of characteristics that can be measured during the interviews. However the selection of characteristics is constrained by practicalities of measuring them. To determine the characteristics the information and process perspective, the technology perspective, and the inter-organizational perspective are used. Using these perspectives should result in a comparable and comprehensive description of the structure.

### 4.4.1 What: Information and Process perspective

What the Supplier Integration System is used for defines from a high process level perspective what it is. But when looking at a more detailed level, the different types of Supplier Integration Systems can support collaborative transaction management and collaborative planning, in various ways. Therefore in this section sub-processes and information components are identified that are expected to be supported or used in some implementations and not in others. Besides process and information types two more characteristics that describe functional aspects are identified.

To determine what sub-processes are supported and what information types are used one has to look at the functionalities of the information system. To describe the functionalities of information systems in detail there are many modeling methodologies. For instance the Unified Modeling Language (UML) is a standard modeling methodology to create various models to describe different aspects of information systems in detail [8]. But this would require too much detail in order to produce additional value and does not provide a reference. The Supply-Chain Operations Reference (SCOR) framework is much more suitable for the purpose here. SCOR was developed by the Supply-Chain Council and provides a crossindustry, standardized process reference model [47]. The Supply-Chain Council (SCC) is an independent, not-for-profit, global corporation with membership open to all companies and organizations. SCOR was designed to help companies to improve their Supply Chain Management. It provides standardized descriptions of management practices, of processes and their relations, of metrics for performance measurement and of best practices. Here it is used to describe and distinguish the sub-processes of collaborative transaction management. In Table 5 the SCOR level 3 source processes are listed. They cover the same activities that fall in the category of the operational purchasing processes as identified by van Weele [3] (added in between brackets), but provide slightly more detail. The transferring process of the SCOR model is left out because it is not relevant here.

Schedule Product	Scheduling and managing the execution of the individual deliveries of product against an
Deliveries	existing contract or forecast. The requirements for product releases are determined based on
(Ordering)	the detailed sourcing plan or other types of product pull signals. This process includes
	(re)ordering and approval.
Receive Product	The process and associated activities of receiving product to contract requirements. This is a
(Expediting)	part of fulfillment or expediting. This process starts with a shipping notification and ends
	with the delivery notification of the receiving stocking location.
Verify Product	The process and actions required determining product conformance to requirements and
(Evaluation)	criteria. This is a part of fulfillment or expediting. Also called quality management.
Authorize	The process of authorizing payments and paying suppliers for product or services. This
Supplier Payment	process includes invoice collection, invoice matching and the issuance of checks.
(Invoicing and	
Follow up)	

 Table 5: SCOR level 3 source processes (with adaptations) [47]

Although the SCOR model does provide planning processes it is not the most suitable in this situation. Collaborative planning processes typically follow a generic cyclical process [48]. In Table 6 the sub-processes are listed. In general these processes are repeated on a regular basis for varying overlapping time-intervals and are often refreshed. This means for instance that there are plans that cover a month or a year ahead that are refined on a weekly basis. During these weekly iterations the entire cycle of sub-processes is repeated. A notable exception is when the VMI concept is applied. Then there is no need to exchange and negotiate plans. The vendor is responsible for the customer's inventory and has the sole responsibility for planning. Local available information should be exchanged in agreed time intervals though, exceptions should be handled and the plans should be executed and evaluated.

Local Domain Planning	The process of developing a plan ("courses of action over specified time periods that represent a projected appropriation of supply resources to meet sourcing requirements" [47]) based on locally available information and the process of gathering relevant local information.			
Plan Exchange	The process and associated activities of exchanging plans and relevant local information. This encompasses sending and receiving local information, the activities of ensuring correct communication and evaluating the received information.			
Negotiation & Exception Handling	The process and associated activities of reaching an agreement on exceptions and committing on a joint plan, based on received plans and other relevant information.			
Execution (of the source processes in Table 5)	The process of converting the agreed plan into the transaction management process and the activities to keep local information up to date.			
Evaluation	Execution performance is measured and evaluated. This could lead to compensation schemes.			

### Table 6: Collaborative Planning processes [48]

Besides the supported business processes the information that is gathered, stored, processed, exchanged and re-used, is distinctive for different Supplier Integration System types too. Information is required to make decisions and to facilitate the various processes. The kind of information listed in Table 7 is based on information that is exchanged in the paper based operational purchasing process and literature on Supply Chain Management [10, 46].

× · · ·	
Inventory position	The amount of inventory of a certain item on stock. The calculations for this
	type of information can vary though. Besides the items on stock, those that are
	in production or already part of finished goods minus the requirements for
	back orders can also be part of the calculation
Point-of-sale data	The amount of actual sales of the customer. This is added to historical POS
	data. This can be used to predict sales forecast of customers.
Forecast	The forecast of orders the customer will place in a certain time-interval
	according to contractual agreements. The customer calculates their
	requirements for a certain time-interval and makes a plan to order these
	requirements. This is the demand forecast for the supplier.
Capacity	The supplier's capacity to deliver certain amounts of items in certain time-
	intervals.
Lead time	The period of time between the order and the delivery at the customer. The
	calculation is
Order	Message from the customer to the supplier with aggregated order-lines for
	required items to be delivered in certain time-intervals according to
	contractual agreements
Order Confirmation	Message from suppliers that confirm orders and commit the supplier to deliver
	according to contractual agreements
Advanced Shipment Notification	Message from the suppliers that precede deliveries, informing the customer of
-	the delivery schedule and all other relevant information that is required to
	receive the items properly
Good receipt confirmation	Message from the customer that the describes the conformance of the delivery
-	to requirements
Invoice	Message from the supplier that aggregates the check of supplied items to be
	paid according to contractual agreements

**Table 7: Information types** 

Besides the information and process types four other aspects are of interest here [27, 31, 38].

System Intelligence	The level of reasoning capabilities the system can offer to support users. This ranges from simple business rules that check messages to advance planning support and intelligent agents that can initiate and complete processes autonomously.
Integration timing and delay (also	The average time required between release of information or any other signal by the senders back end system and the moment that the receivers back end systems receive the message.
called Speed)	Real-time $(0 - 0.1 h)$ Near real time $(0.1 - 1 h)$ Batched $(1 - 24 h)$
Volume	% of transactional document flow supported by the Supplier Integration System
Diversity (also	% of transactional document types supported by the Supplier Integration System
called range)	

**Table 8: Functional aspects** 

### 4.4.2 How: Technology perspective

The second feature that distinguishes Supplier Integration System structures is determined by the underlying technology that is used. There has been a history of developments in the field that resulted in a considerable amount of technologies that enable supply chain integration. Even from a purely technological point of view there is no one best way. But it is sufficient to realize that in practice there are many different technologies available and used, because of, for instance, legacy systems in established companies. In Table 9 characteristics are listed based on the characteristics of networked systems in general (also see Table 4) and based on the characteristics suggested by van Hillegersberg [27].

Process & Data Standard	The process & data standard used for the messages and joint processes. Open vs. closed or less open standards. These standards can be Proprietary -, Industry - or Universal standards. They can be based on older EDI standards or based on newer XML standards [32].	
Governance structure	Management and control of the Supplier Integration System: this can Centralized vs. decentralized, and include the presence and role of a third party. Also the ownership of the system is an important part of this	
Process Architecture	Business process management and workflow management is based on a Hub-and-spoke or federated architecture. Hub-and-spoke means that the system manages the processes at the central level. Federated means that the system connects local systems that do the process management itself.	
Network type	The network technology used. Dedicated communication channels (ISDN), Value-added- networks (VAN) or the Internet.	
Technology specificity	The technology used for the system can be either proprietary ("home-made") and very specific or broad based to varying degrees. The latter depends on the diffusion of the technology and degree of customization of a broad based technology. (this influences agility)	

Table 9: Technological and structural aspects

### 4.4.3 Context: Inter-Organizational perspective

From an inter-organizational perspective the system is larger than the application itself and includes the actors and their relations and economic properties. These characteristics are based on the characteristics of networked systems identified in literature to distinguish e-procurement types or networked systems in general (also see Table 4). These characteristics are descriptive of the situation but are in some cases also expected to have a strong impact on many of the information, process and technological aspects of the Supplier Integration System structure [38].

Breadth (also called reach)	% of trading partners linked by the Supplier Integration System
Supplier type	Size, specific knowledge, available resources
Product types in scope	Degree of Complexity (maturity, technical complexity and engineering effort), Specificity (level of customization, amount of alternatives feasible), Single unit costs, Impact on cost price of end product, contribution to selling power of the end-product
Relationship type	Arms' length vs. exclusive partnership, level of trust,
Relationship asymmetry	level of required coercion
Order type / Contract horizon	Spot vs. systematic orders / Short vs. Long term contracts
Pricing mechanism	Fixed pricing, auctioning, bargaining, brokerage, discounting

**Table 10: Inter-organizational characteristics** 

### 4.5 Conclusion

This chapter explained what the Supplier Integration System is, by listing general characteristics and contrasting them with other e-procurement types. This is used to define the subject and the scope of the study. In Table 11 characteristics, which can be considered general Supplier Integration System features are summarized. These features combined distinguish them from other e-procurement types. The other characteristics listed in Table 4 cannot be assumed as general characteristics. These remaining characteristics can vary for different Supplier Integration Systems.

Perspective	Classification Characteristic	Supplier Integration System feature
Information and	Process types	Supporting the Operational purchasing process with
process		collaborative transaction management and collaborative
		planning.
Inter-organizational	Product type	Direct items; defined as the items (products and materials) used
		in the production process (e.g. catalysts) or that end up in the
		final-product. These are Bill-of-Material items.
Inter-organizational	Collaboration /	Vertical supply chain collaboration (1:M); one buyer collaborates
	Network type	with its many suppliers. Suppliers do not have a direct
		relationship with other suppliers concerning the system. There is
		a sequential interdependency.
Inter-organizational	Power distribution /	Buyer-centric: this study focuses on the buying organization.
	Control	They are selected for their size and capabilities. They are all
		expected to be the most dominant actor, although in various
		degrees.

Table 11: General characteristics of the Supplier Integration System

In the communication with (potential) respondents the general characteristics are used as a reference for the interviews. The more detailed and distinctive characteristics are then used to determine, with the respondents, what the characteristics of their particular implementation are. The characteristics listed in Section 4.4 are used during the interviews as a checklist. In this way a description of the structure is obtained in a comparable way. In Chapter 5 factors affecting adoption process and its resulting structure are explained. It should be noted though, that many of the characteristics listed here are probably affecting each other too. As was for instance suggested in Section 2.2.1, breadth (reach) of the system is positively impacted by the newer e-business technologies.

### 5 The adoption process of Supplier Integration Systems

The diversity of Supplier Integration System structures was described in Chapter 4. The Supplier Integration System was explained as a class of inter-organizational systems. A checklist of characteristics was created to describe and distinguish different structures. Besides recognizing a Supplier Integration System this enabled a structured and comparable description of Supplier Integration Systems at the studied companies. The structure of a Supplier Integration System is the result of the adoption process model used in this study, as the model includes the motivations, enablers, inhibitors and the organizational actions leading to the adoption or rejection of a system. Therefore this section will explain the various theories of adoption that form the basis of this model and how these theories relate to the structure of Supplier Integration Systems. This chapter will answer the second sub-question about the adoption process and partially the third sub-question about the relation. The questions 2a, 2b and 3a are answered:

# **2:** Adoption Process: How can the Supplier Integration System adoption process (best) be studied?

- *a.* What are the different perspectives on adoption processes of inter-organizational systems?
- b. What are the contributions and shortcomings of these approaches?

# **3:** Relation: What is the relation between the adoption process and Supplier Integration System structures?

a. What theories explain the relation between the adoption process and Supplier Integration System structures?

In order to answer the main research question this study aims to enhance the adoption process model of Kurnia and Johnston [24] (also see Section 3.1.2). Therefore the processual approach to adoption is also followed in this study. In this chapter this processual view of adoption is explained. It is also explained how this model relates to and complements other theories of adoption. Especially with regards to the factor model of adoption as it is an extension of these models. To make these differences explicit, the taxonomy of adoption theories is discussed first as was done by Kurnia and Johnston [24].

Following this taxonomy the theories based on the most common perspectives, the factor approach, are discussed. These studies have resulted in models explaining what factors are necessary for adoption of Supplier Integration Systems. They include enabler, inhibitors and motivations to adoption in a technology–organization–environment (TOE) framework. The discussion of these models will cover both the merits of these models and the shortcomings.

The processual approach to IOS adoption is explained next and how this approach overcomes some of the shortcomings of the factor approach.

Finally the last shortcoming, which is not tackled by the process model, is discussed. Many studies, including the process model of Kurnia and Johnston, assume a dichotomous outcome (adoption or rejection), while in reality there are many different structures and therefore multiple outcomes possible. As was argued in Chapter 3 this is an oversimplification that cannot be ignored as different conditions and objectives require different Supplier Integration System structures. Some theories attempt to overcome this though, by studying the relationship between certain factors and some aspects of the structure of IOS. These theories explain what determines the structure of IOS to some extent and are therefore discussed next.

The results of this chapter are used in the case interviews to guide the story telling of the adoption process. Following the assumptions of the processual approach the respondents are asked about what happened and what was decided during their adoption process.

### 5.1 Perspectives on IOS adoption

As was discussed in Chapter 2, the initial innovation of IOS was followed by many smaller innovations, of which the Internet was a major one. Innovation "refers both to the output and the process of arriving at a technological feasible solution to a problem triggered by a technological opportunity or customer need"(p68 [13]). Adoption is a demand side concept that entails the acceptance of an innovation. This assumes that the person or organization is capable of adoption of the innovation. The adoption process starts when the organization becomes aware of the innovation and ends when the innovation is implemented or abandoned.

The concept of adoption has been studied extensively. This resulted in many different perspectives on the concept and theory of adoption. In this study the perspectives advocated by Kurnia and Johnston are used in what they call the processual view of IOS adoption, to complement the factor approach used in most studies [24]. They used the taxonomy of Markus and Robey [49] to describe the perspectives of other studies and their perspective. The taxonomy distinguishes three dimensions of causal structure of information technology adoption: causal agency, logical structure, and level of analysis.

Causal agency refers to beliefs about the nature of causality and has three perspectives:

- 1. *Technological imperative*: The technology and other external factors determines or strongly constraints individual or organizational behavior including adoption. Also called "situational control" perspective.
- 2. Organizational imperative: Rational choice consistent with preferences and goals determines individual or organizational behavior including technology adoption. Human actors have almost unconstraint control over the technology adoption. Also called "rational actor" perspective
- 3. *Emergent perspective*: individual or organizational behavior including technology adoption emerges unpredictably from the interaction of people, events and external factors

**Logical structure** is about the temporal aspect of the theory and the relationships between predicting variables and the outcomes. It has two perspectives:

- 1. *Variance theories*: These theories assume that certain variables measured at a certain time are necessary and sufficient to determine outcomes
- 2. *Process theories*: These theories assume that a set of sequential conditions is necessary over time to arrive at certain discrete outcomes. But even when all conditions are met outcomes may not occur, because of dynamic interactions between technology and actors.

**Level of analysis** refers to the entities about which the theory poses concepts and relations. With respect to technology adoption this dimension ranges from the highest macro-level of societies to the lowest micro-level of individuals.

### 5.2 The factor approach to IOS adoption

Many empirical studies of IOS adoption in organizations have used the factor approach. In terms of the Markus and Robey taxonomy; they assume the technological imperative, develop variance theories and study at the organizational level of analysis. The models studied, using the factor approach, are often based on the technology–organization–environment (TOE) framework [24, 26], developed by Tornatzky and Fleischer [50], explaining general technological innovation adoption by firms. The approach used in these studies, such as the EDI adoption studies by Iacovou et al. [28] and Chwelos et al. [51], assumes that factors in the technological, organizational or environmental context determine and sufficiently explain the

process of organizational adoption. This simple causal structure is empirically tested with surveys measuring the existence of these factors (the predicting variables) in relation to the dependent variable of IOS adoption. There have been enough statistically reliable studies to accept the reliability of the resulting factor model. The result is a factor model with a considerable list of factors in the three categories of the TOE framework. They include enablers and inhibitors of adoption and they include the implicit and explicit motivations of companies to adopt or reject the system.

**Technology**: The factors in the technology categories are about the perceived nature of the technology including existing technologies inside the firm, as well as the pool of available technologies in the market. In this case it pertains to the perceived characteristics of EDI and e-business technologies. The perceived costs, disadvantages and benefits of the new technology are measured. It is intuitively clear that when decision makers have to decide over the adoption of a new technology that they look at what the advantages and disadvantages of the new technology are.

**Organization:** The factors in the organization category are typically about the capabilities of the organization and other internal organizational characteristics such as firm size. It is intuitively clear that decision makers look at their own organization and assess whether or not they are capable of successfully implementing a new technology.

**Environment** The environment category contains the external factors, such as market pressure or government interference. Decision makers constantly try to adapt their organization to optimally fit in their environment, this includes perhaps adopting a new technology when the environment makes this necessary to survive as a healthy company.

In Figure 9 a TOE factor model that combines results of several empirical studies is depicted. It is an appealing model because it can be readily translated into a set of guidelines for practitioners. Take for instance the factor of top management commitment. This indicates that without top management commitment it is less likely that the organization will be able to adopt an IOS. Using the factors in this way as a checklist, companies can asses whether or not they are ready to adopt an IOS.



Figure 9: TOE Factor model of IOS adoption [24]

But the model has three shortcomings:

- 1. The list of factors is probably not complete. Although it lists a considerable number of factors, other studies have found other factors, such as firm size or firm scope, having important influences on IOS adoption [26]. Refinement and extensions therefore still remain necessary. This is also suggested by Kurnia and Johnston, as they do not provide a conclusive list of factors [24].
- 2. The underlying assumptions of the factor approach cause several shortcomings. Although the factor approach produces a set of necessary factors, these factors are not sufficient to account for the richness of adoption experiences [24]. The factor model neglects the dynamics of an IOS adoption process in an inter-organizational context.
- 3. The assumption of a dichotomous outcome is an invalid simplification of reality. Assuming that an organization either adopts e-business or does not, was useful for studying the rise of e-business. It helped managers to decide whether or not to go along with the new technologies. But with the ongoing proliferation of Supplier Integration Systems the question shifts from the adoption decision to the decision what structure is most suited. As was shown in Chapter 4 there are many different structures possible. As was argued in Chapter 3 this is an oversimplification that cannot be ignored as different conditions and objectives require different Supplier Integration System structures.

Because of these shortcomings the factor model is not suited to answer the main research question. The first shortcoming is not caused by fundamental flaws of the factor model but simply because of the large number of possible factors. The cases studies can only confirm factors, add richness and possibly can discover overlooked factors. The second shortcoming is about the underlying assumptions. The processual approach, explained in section 5.3, is used to reduce these fundamental shortcomings of the factor approach. But the process model of Kurnia and Johnston [24] does not mend the third shortcoming, as was already explained in section 3.1.2. Some theories that treat this shortcoming do exist though. Therefore these theories about the impact of certain factors on some aspects of the structure of Supplier Integration Systems are discussed in section 5.4. In the case studies specific attention is paid to these factors in relation to the structure of the Supplier Integration System.

### 5.3 The processual approach to IOS adoption

In this study the perspectives advocated by Kurnia and Johnston are used. They call their view the "processual view" of IOS adoption [24]. Contrary towards many studies using the factor approach their processual view advocates studying IOS on firm / inter-firm level using the emergent perspective and a process logical structure.

**Firm / inter-firm level of analysis:** Theories explaining the adoption of IOS should cover the focal company and its partners. This is important because the inter-organizational environment is inherently part of the IOS. The relationship with the trading partner illustrates this. The factor approach cluster factors covering the relationship with the trading partner in the environment category (e.g. trust). This implies that the organization has no influence on the nature of the relationship while in reality the relationship is off course affected by past interactions, attitudes and mutual agreements. The relationship is also affected by the behavior of the partner. Even the adoption process of IOS affects the relationship itself. In other words: organizational actions impact the relationship. Therefore the firm/inter-firm perspective is the appropriate level of analysis. This results in a distinction between environmental forces the organization cannot affect (external factors) and supply chain /
industry factors which become internal factors and are part of the organizational actions. In this way the complex interaction dynamics of the adoption process can be understood better.

**Process logical structure:** The study should have a longitudinal character taking into account the process of adoption as a long term project. During this process many conditions have to be met at given points in time. This is an important notion, because in reality it takes time and involves complex inter-organizational interactions to adopt IOS. The factor model is appropriate to identify the necessary conditions for adoption but not sufficient to explain this process. When looking at the adoption process over a longer period of time it becomes clear that the significance and nature of factors change over time.

**Emergent perspective of causal agency:** The emergent perspective is important because typically companies are not in total control of their environment, but are not helpless victims either. Organizational actions influence the environment, the supply chain, the perceived nature of the technology and its own organizational capabilities. And in turn these factors affect organizational actions.

Kurnia and Johnston's processual approach resulted in a more complex model, depicted in Figure 10, which is better able to explain the complex dynamics of the adoption process. The TOE framework is extended with the "supply chain / industry structure" category representing the inter-organizational domain and extra arrows to represent the emergent causal structure.



Figure 10: Second order model of IOS adoption (with some adaptations<sup>2</sup> [24])

#### 5.4 Adoption and the structure of Supplier Integration Systems

This section focuses on the relation between factors, organizational actions and the outcome of adoption and use IOS, since this study aims to extend the second order model in Figure 10 with a more diverse outcome. As was seen in Chapter 4 many different types of IOS exist and also the structure of Supplier Integration Systems is pretty diverse. Building on the processual approach the Supplier Integration System structure is a result of the complex

<sup>&</sup>lt;sup>2</sup> The blue output box was called "Adoption of ECR" in the original model. ECR was used as an example of IOS.

dynamics of the adoption process. These are the same interaction dynamics between factors and organizational actions described in section 5.3. The structure of the Supplier Integration System must be compatible with relevant organizational practices and policies (which are part of the organizational processes) [52] Like in any IT investment the strategic alignment of the technology and the organization and its environment is essential to realize value from these investments [53]. As organizations try to realize this fit, they try to influence the perceived nature of the technology, the organizational capabilities and the supply chain, like modeled in Figure 10 and at the same time tailor the structure of their new Supplier Integration System. Although this study is exploring relative new terrain when studying the adoption process in relation to its diverse outcome, a number of aspects of the outcome have been studied before. The following sections will treat some aspects studied in prior empirical research in relation to characteristics of a Supplier Integration System.

#### 5.4.1 Processes and Information sharing

The adoption process leads to different outcomes concerning the processes that are integrated and the information that is shared. The extent to which processes are integrated and information is shared depends on many of the conditions in the second order model (Figure 10). Two critical factors in the adoption and use of IOS are power and trust [30]. Further empirical research showed that relational factors significantly impact the diversity of integrated processes and shared information [54]. It showed that the transaction channel climate and the partner persuasion method have significant impacts.

The transaction channel climate describes the atmosphere of a relationship between business partners. It can be expressed in terms of the cooperation level in a buyer-supplier relationship. A cooperative channel climate has more diverse integrated processes and shares more information. In the case of the Supplier Integration System this would suggest that in a cooperative climate more of the processes are integrated and that it is more likely that sensitive information is shared. Three determinants impact the channel climate. The level of trust that the partner's future actions will be beneficial (or at least not harmful) and the level of asset specificity, meaning that the partner has made relationship specific investments, positively impact the level of cooperation. Uncertainty about partner's future actions is negatively associated with the level of cooperation.

The partner persuasion method is about how the partner is treated by the initiator of the IOS when they try to connect their partners. Two strategies are distinguished. These are exercising power (the stick) and reciprocal investments (the carrot). While power (either explicitly exercised or not exercised) does force partners to adopt the IOS it would lead to low usage in terms of less diversity of processes. The reciprocal investments in the form of IOS related support, especially in a cooperative climate, often lead to a much more diverse amount of integrated processes. Therefore the more elaborate IOS are probably the result of more reciprocal investments and a generally cooperative atmosphere.

#### 5.4.2 Standards

An important aspect of Supplier Integration Systems is what process and data standard is used. The more recent standards based on XML are considered more open standards and increasingly important. But older proprietary and less open standards, such as classic EDI, are still used. In order to better understand the process of standards diffusion, firms' migration from proprietary or less-open standards to open standards was studied empirically [32]. This study, by Zhu et al., showed that network effects and path dependency have significant impacts.

**Network effects** theory poses that benefits derived from a network technology is positively associated with the size of the network. When more peers adopt a certain standard (the network technology) the network size increases. Through network effects the benefits for each adopter will then increase because more partners are using the same standard allowing the adopter to share information and integrate processes with more partners. Expected benefits are off course significant drivers of adoption. But there is also a more direct network effect as peers and trading communities try to put pressure companies to adopt a certain standard because this also increases their own expected benefits. Peer adoption and trading community influence are therefore determinants for the strength of network effects.

**Path dependency** is the notion that a firm's ability and incentive to adopt a newer technology are largely a function of its level of related experience with prior technologies. Path dependency manifests itself both through the experiences of people and through past investments. Past investments in the organizational infrastructure may be lost when switching to a new technology. Classic EDI users often have made specific investments to create their inter-firm linkages. When adopting a new standard they would loose these investments and thereby incur additional switching costs besides the normal adoption costs. Switching costs are also consistent with lower incremental value of the new standard. At the same time, though, having a suitable infrastructure may reduce the costs of adoption when some of the hardware or software can be reused. Past experiences also have a twofold effect. They reduce adoption costs because the organization has learned from their past experience with similar technology and the organization is therefore better able to cope with managerial complexities. But these past experiences also tend to make the organization more cautious as they give the organization a better understanding of the adoption costs and potential benefits involved. New adopters may be unaware of the intangible costs and therefore underestimate the adoption costs or fear to fall behind technologically and therefore overestimate benefits.

#### 5.5 Conclusion

This chapter explained how this study will treat the process of adoption in relation to its structure. The case studies are examined using the processual approach. This means that attention is paid, not only to the factors, but also to the historical development of the Supplier Integration System into the structure it is today. Respondents are asked to comment on choices made during the adoption process. What were the factors in these decisions, what were the actions to influence these factors and how did this dynamic process roll out. Were there any unexpected events or results? How is the final outcome perceived? And what are the future developments and courses of action? The factors listed in the second order model are used as a guideline for the interview. Specific attention is paid to the theories explaining some structural aspects of the Supplier Integration System, treated in this chapter.

# 6 Case Analysis

This chapter covers the empirical part of this study. Eleven case studies at large global manufacturing firms are performed. In Chapter 3 the study method and selection process was explained. In Chapter 4 and 5 the theory used as guidelines for the cases was described. In this chapter the results from these case studies are analyzed. Structure, adoption process and the relation between the two are treated here. This means that the remaining sub-questions (1d, 3b and 3c) are answered:

#### 1. Structure: What are the various Supplier Integration System Structures?

d. What Supplier Integration System structures can be observed in practice?

**3.** Relation: What is the relation between the adoption process and Supplier Integration System structures?

- b. What adoption processes can be observed in practice?
- c. What explanations of the specific adoption process and its resulting Supplier Integration System structure can be observed in practice?

In the interviews the structure of the Supplier Integration Systems appeared to be the most suitable guideline throughout the interview. The impact of the adoption process (including its decisions) on specific aspects of the structure was discussed when these aspects were talked about. When a good description of the structure was established the focus shifted to an explanation of the decisions made. Also the perception its perceived benefits and costs, risks and occurring problems came up.

This chapter will follow the same organization as used in the interviews. The structure of the Supplier Integration Systems is leading. Section 6.1 will therefore describe the observed basic structures first. This in part answers sub-question 1d. Section 6.2 up to section 6.6 will go into the details of these answers. These sections also bring up the explanations from the adoption process for these aspects of the structure. Therefore these sections also answer question 3b and 3c. The adoption process is discussed in Section 6.7, providing more answers to sub-question 3b and 3c.

#### 6.1 Basic structures

In academic and business literature terminology was used in a confusing manner (see Chapter 3 and 4). Also in the case interviews terminology appeared confusing. Terms required explanation and a systematic approach to cover the various aspects comprehensively. In Chapter 4 these various aspects of the structure were explained. In the interviews these tables were used as a checklist to systematically describe all the aspects of the system resulting in comprehensive descriptions of the Supplier Integration System structures. In these descriptions three fixed combinations of characteristics of the structure could be observed. These fixed combinations of characteristics are therefore called the basic structures. The three observed basic Supplier Integration System structures are: the system-to-system structure, the VAN structure and the supplier portal structure. These three basic structures have distinctive characteristics when compared to each other. Of course individual systems within these basic categories also have distinctive features.

The first three sections explain these basic structures. The basic structures are applied to different integration settings. The fourth section explains the basic integration focus. This

integration focus, describing the position of the system in the value chain, was not theorized in Chapter 4, but in the interviews it appeared to be an important aspect of the context. Finally the case observations of the basic structures are listed in the final section.

#### 6.1.1 System-to-system structure

The system-to-system structure is the oldest Supplier Integration System structure. Traditional EDI systems are the main example. But also modern integration systems based on XML standards such as RosettaNet, fall in this category. They are also called virtual integrated supply chains [39].

The most important common feature is that there is a direct system-to-system link that integrates partner back end systems. Both the supplier and the buyer arrange their own integration software that provides the interface. There is agreement about a common data and process standard. The integration software transforms the data from back end systems into this common data standard according to the processes agreed. The connectivity is locally arranged using a network technology such as a dedicated phone line or the internet. The process architecture is federated. The Supplier Integration System can basically be viewed as a messaging system between otherwise separate systems.

Figure 11 shows a schematic representation of this structure. This figure shows 1 buyer and three suppliers to keep the representation simple. In reality the basic structure accommodates varying networks with one or more suppliers and one ore more buyers.



Figure 11: System-to-system structure

The system-to-system structure offers the possibility of a very high level of integration with the integrated partner. Common processes and information sharing can be completely automated. The main drawback is that with each partner a common data and process standard must be agreed. When there are no industry standards, or when there are many suppliers with many differences this proves to be very costly. Even when using open standards, practitioners face high implementation costs. In a number of cases, such as FMCG and Electronic4, this only proved worthwhile with important suppliers. Important suppliers are suppliers with high volumes, high order/delivery frequencies, and long term relationships. The organizational and IT capabilities of the partner and the willingness to cooperate also played an important role. Within this structure for each partner a separate business case is often required to support the decision to integrate them.

#### 6.1.2 VAN structure

The name of this basic structure is based on the value-added network (VAN) service that appeared soon after the introduction of traditional EDI. These value-added networks are provided by integration service providers. These are intermediaries that act as a hub of information and thereby decrease the number of required connections in large networks with many buyers and many suppliers. Traditionally they provided traditional EDI services but today also modern technologies are used and a wide range of extra services can be provided.

The basic common feature of the VAN structure is a hub-and-spoke architecture using an intermediary entity. Two typical case examples are "e-invoicing" and the "e-hub logistics service provider". In the case of e-invoicing the invoicing process is supported by an external e-hub service provider. Suppliers' financial systems as well as the buyers' systems are integrated with this e-hub enabling the automated processing of invoices. According to a recent survey this is a key topic in many companies [55]. The logistics service provider example is an even more elaborate service supporting the entire planning and transaction management process.

The VAN structure encompasses what is by some viewed as a form of electronic marketplace in a business to business setting aimed at vertical collaboration [37]. The e-marketplace is in this view a platform or hub supporting integration in hierarchies. It should not be confused with b2b hubs for market-mediated interactions.

Figure 12 shows a schematic representation of the VAN structure. It shows one buyer, three suppliers and an integration service provider in between. In reality the structure encompasses varying network structures with one or more suppliers, one or more buyers and one integration service provider. Also in the case of e-invoicing the distinction between suppliers of direct and indirect items disappears.



Figure 12: VAN structure

The VAN structure offers possibilities of high integration levels and outsourcing services, such as supplier development. An external integration service provider actively contributes to increase the network of integrated partners, as this specialized intermediary seeks new customers. Due to the network effects (see Section 5.4.2) it is inclined to do this. In this way, while only having to integrate with one partner (the integration service provider), a large network can be attained. Using the VAN structure can also help to show "one face" to partners. This was the idea behind the failed integration project of Electronics1 to use a VAN

structure with an internal (centralized) integration service provider. This centralized hub was to be the intermediary between the various divisions/departments and its suppliers. It proved to be very difficult for this intermediary to integrate with the many different actors on both sides. The diversity of these actors and their wishes were too great causing the project to fail. External integration service providers have more experience in integrating various actors and already have integrated networks. This is appreciated as the Electronics4 response illustrated: "managing another system is not our core business". The drawbacks are that one becomes dependent on these integration service providers and that they are costly and are in some cases not developed enough to suit the specific wishes. The VAN structure with external suppliers is not much adopted yet, by the case companies (only FMCG has a running system with a VAN structure with an external intermediary), but it is on the agenda of others too (e.g. Electronic4).

#### 6.1.3 Supplier Portal structure

The supplier portal structure is the simplest of the basic structures and has been called a core supply chain business model before [45]. In this structure suppliers can log into secured websites that provide customized information and access to online applications. This portal website is integrated with the enterprise systems of the buying company. Because supplier back end systems are not integrated, employees of the supplier have to manually process and transform the information.

Figure 13 shows a schematic representation of the supplier portal structure. The structure can easily be extended to give access to more suppliers, requiring only an internet connection. At the same time suppliers can end up facing a portal for each of their customers.



Figure 13: Supplier portal structure

The supplier portal structure is the easiest and most accessible solution, but at the same time does not provide end-to-end integration. The Supplier Portal structure does not require the coordination efforts to align data and process standards with partners such as in the other two structures. The partners do not need sophisticated organizational and IT capabilities to use the portal. The only need an internet connection and need to learn how to use the portal and its applications. This is relative cheap and easy to implement. But the drawback is that it does not provide end-to-end integration and thus still requires manual operations to be performed. Also the portal costs the supplier extra work because they need to adjust their normal operation routine to work with it and in some cases can end up facing various different portals with different processes for multiple customers.

#### 6.1.4 Integration focus

It appeared in the case studies that the three basic structures described in the last three sections appear in different integration settings. This setting is an aspect of the context that describes the position of the system in the supply chain. This position is the focus of integration by the integration system. Two aspects describe the basic integration focus.

The first is the distinction between so called internal suppliers and external suppliers. Many of the studied corporations own themselves large parts of the supply chain. This means that supplier plants are owned by the same company as the buying plant, hence so called internal suppliers. For instance in the case of Truck2 the truck assembly plant procures its materials and components at various external suppliers, but the most important components such as the engines are obtained from internal suppliers. Truck2 manufactures the engines themselves. In the case of Packaging1 their own paper mills produce for their own corrugated and card-board plants. Here it gets even more complicated as the paper mills also sell to external customers and the corrugated- and card-board plants also buy from external paper mills.

In literature there is a very clear distinction between Enterprise Application Integration (EAI) and business to business (B2B) integration, but it appears that the same systems are used for both internal and external integration. Indeed in literature it is acknowledged that the same technologies can be applied for both types of integration [8]. It is also important to notice that integration of the internal supply chain gets precedence over integration of external suppliers. The reasons mentioned for this are that it is perceived to be easier to integrate internal suppliers, that internal suppliers are considered more important suppliers (resulting in more benefits from integration) and that it is considered a pre-condition before one can integrate with partners.

The second aspect describing the integration focus is the basic type of actor. It is either a plant (a producing actor) or a distribution unit (a sales unit). This results in an integration focus that is either plant-to-plant or plant-to-distribution. In the case of Truck1 the plant-to-plant integration focus is between producers of components and the assembly plant. The plant-to-distribution integration focus appears between plants and sales agents in the Pharma2 case, and appears between contract manufacturers and the OEM in the case of Electronic3.

The trend of increasingly using contract manufacturers (especially in the electronics industry) appears to shift the integration focus for the studied companies from plant-to-plant towards plant-to-distribution. The plant-to-distribution integration focus is easier because there are generally less suppliers, a smaller diversity of items and a reduced bullwhip effect (because one is closer to the end-customer).

#### 6.1.5 Governance structures and process architecture

Each of the basic structures has a distinct governance structure and process architecture. Following Table 9 three aspects of the governance structures and process architecture were inquired.

The first aspect pertains to the *ownership* of the system. Most of the observed systems are owned by the focal firm. Only in the cases where a VAN structure with an external (third party) intermediary is used the Supplier Integration System is owned by one other then the focal company: the third party intermediary.

The *management* of the system is a more complex aspect of the governance structure. Usually it is centralized at the divisional or company level. But then also at the plant level operational management tasks related to the system occur. In some of the cases there is an IT outsourcing partner. The management of this partner with respect to the Supplier Integration System is usually centralized. In case of the VAN structure with an external intermediary many of the management tasks are taken care of by the third party. Examples are the development and

encouragement of suppliers to cooperate in the integration system and the management of the IT.

The *process architecture* is directly related to the basic structure. The system-to-system structure typically has a federated process architecture. While the VAN structure and the Portal structure typically have a hub-and-spoke architecture.

#### 6.1.6 Case observations

The case observations of the basic structure, integration focus and the governance structure are listed in Table 12, Table 13 and Table 14. One can see that the three basic structures are used in a complimentary way.

Company	Description of the studied integration system (s)	Integration focus			
Pharma1	1. System-to-system structure: Integration with Contract Manufacturers is supported with interfaces. They have just recently become independent (since 2006). Integration with suppliers is planned (The business case is approved). Internally the supply chain is fully integrated with one enterprise system.	Plant-to- Plant	External		
Pharma2	<i>1. VAN structure</i> : Integration between sales affiliates (local companies) and plants is supported with advanced planning and e-ordering. There is an internal hub since 2005 (idea started in 2001).	Plant-to- distribution	Internal		
Packaging1	1. System-to-system structure: The system integration / VMI system on the e-business platform supports the integration of internal and external paper mills with internal and external corrugated and card board plants since 2001.	Plant-to- plant	Mostly Internal		
Packaging2	1. System-to-system structure: System integration between internal paper mills and mostly internal and some external cardboard plants is supported based on VMI (since a few years)	Plant-to- plant	Mostly Internal		
Truck1	<ol> <li>System-to-system structure: The classic EDI system (started in 1987) supports integration between mostly external suppliers and the assembly plant</li> <li>Supplier portal structure: Additionally there is web-EDI (supplier portal) for smaller partners (with mostly the same functionalities) due life in 2008</li> </ol>	Plant-to- plant	Mostly external		
Truck2	<ol> <li>System-to-system structure: The classic EDI system (started in the early 90s) supports integration between mostly external suppliers and the assembly plant.</li> <li>Supplier portal structure: Additionally there is web-EDI (supplier portal) for smaller partners (with mostly the same functionalities) since a few years</li> </ol>	Plant-to- plant	Mostly external		
Electronic1	System-to-system, Supplier portal and VAN structure: The current integration efforts on a divisional level with suppliers and Contract Manufacturers are discussed. This includes some EDI and supplier portals. E-faxes are also used a lot. The failed project to integrate all external suppliers and contract manufacturers with all plants through one firm wide e-business hub (VAN structure with internal intermediary) is also discussed.	Plant-to- plant / distribution	External		
Electronic2	1. System-to-system structure: Integration of transactional process between plants and sales affiliates is supported. The Dutch situation is investigated.(running for more then 10 years)	Plant-to- distribution	Internal		
Electronic3	<ol> <li>System-to-system structure: Currently there are some older EDI connections, but mostly paper based communications</li> <li>VAN structure: A new integration system is in <i>development</i>. This should integrate suppliers and contract manufacturers with a logistics service provider and the focal company. (focus of the interview)</li> </ol>	Plant-to- plant / distribution	External		
Electronic4	1. System-to-system structure: There is a peer-to-peer integration system (RosettaNet based) with Contract manufacturers since the late 90s 2. VAN structure: E-invoicing (in <i>development</i> ) with all supplier types 3. Supplier portal structure: There is a Supplier Portal with suppliers of direct items. (since +-2003)	Plant-to- plant / distribution	External		

 Table 12: Basic structure and integration focus (part 1)

Company	Description of the studied integration system (s)	Integration focus				
FMCG	1. System-to-system structure: EDI system: to support the VMI concept	Plant-to-	External			
	with two important suppliers since +- 2001	plant /				
	2. VAN structure: E-Invoicing: for suppliers of both indirect and directs.	distribution				
	An external service provider offers integration services for the invoicing					
	process.(since 2007 (development started in 2006)					
	3. Supplier portal structure: Supplier Portal: website where about 100 of					
	the larger suppliers can log in and download information and can commit					
	to forecasts /orders (since +-2003)					

 Table 13: Basic structure and integration focus (part 2)

	Governance structure									
Company	Ownership by	Management	Process Architecture							
Pharma1	Focal company	There is an outsourcing partner responsible for IT management. At the plant level the systems are coordinated (including supplier relations)	Federated							
Pharma2	Focal company	Centralized IT department since it concerns mostly enterprise integration	Hub-and-spoke							
Packaging1	Focal company	Centralized e-business platform management	Federated							
Packaging2	Focal company	Centralized IT department	Federated							
Truck1	Focal company	Centralized at the plant level. Since production is for a large part concentrated at this plant	Federated (EDI) and Hub-and-spoke (portal)							
Truck2	Focal company	Partially centralized and partially plant level. There is an IT and a purchasing service at the organizational level, but at the plant level logistical aspects and operations are managed.	Federated (EDI) and Hub-and-spoke (portal)							
Electronic1	Focal company	Centralized project failed because the requirements where to different. Currently the divisions manage their own integration projects	Federated (EDI) and Hub-and-spoke (portal and failed hub)							
Electronic2	Focal company	Centralized at the global level. The country level is responsible for operations	Federated							
Electronic3	External service provider	The logistics service provider will be responsible for the system and will manage the relations together with the focal company	Hub-and-spoke							
Electronic4	Portal and P2P: focal company e-invoice: external service provider	Portal and P2P: focal company e-invoice: external service provider	P2P is federated Portal and e-invoice use a hub-and-spoke architecture							
FMCG	Portal and EDI: focal company e-invoice: external service provider	Portal and EDI: focal company e-invoice: external service provider	EDI is federated Portal and e-invoice use a hub-and-spoke architecture							

 Table 14: Governance structure

The case observations can be summarized by the following:

- The system-to-system structure is most difficult to implement but enables a high degree of integration.
- The VAN-structure can be combined with outsourcing services by an integration service provider but makes one dependent on a sometimes costly and not always competent service provider.
- The supplier portal structure is the most accessible structure but only provides partial integration, thus still requiring manual tasks.
- The three basic structures have distinct governance structures and are used in a complimentary way for both internal and external integration.

#### 6.2 Breadth and volume

Following Table 8 and Table 10, breadth and volume are, besides diversity (see section 6.3), important indicators of Supplier Integration System usage [56]. They describe the "size" of the system. Generally it is considered that a higher breadth and volume is more beneficial for the owner of the system. In this section first the evolving nature is described, then some explanations are discussed and finally the case observations are presented.

#### 6.2.1 Evolving size

In the observed cases during the adoption process the size of the system gradually increased up to the current levels. Especially from higher volumes most gains are expected. Therefore the more important suppliers are often approached first. Important suppliers are suppliers with high order/delivery volumes and frequencies, and long term relationships. Contract manufacturers are often among these most important suppliers. A standard implementation strategy is therefore to integrate the most important suppliers within a system-to-system structure first. When with these suppliers the integration is successful, then the breadth of the system is gradually increased to include the less important suppliers. At a certain point it is not worthwhile anymore (or possible) to integrate more suppliers within the system-to-system structure. Within this structure for each partner a separate business case is often required to support the decision to integrate them. Then the more accessible and easier Supplier portal structure comes in the picture to reach these smaller and less sophisticated suppliers

#### 6.2.2 Achieving higher breadth and volume

It is striking that only the truck companies achieve a high breadth and volume. Compared to these two, the other companies all have a relative low breadth and volume. A logical explanation of this is that the truck companies have a much longer history of using the same Supplier Integration System. It takes time to integrate your partners in your system and achieving a high breadth and volume is something that requires large efforts and long term commitments. Newer technologies, based on open standards or the internet, are not the reason as the truck companies mainly rely on the mature classical EDI technologies. They do use web-EDI (Supplier Portal structure) in a complimentary way to reach smaller and less sophisticated suppliers. Most truck suppliers are integrated with classical EDI (system-to-system structure).

In achieving higher breadth and volume the cooperation of partners is considered essential in every case. Two tactics (or a combination of the two) have been used in the cases to get this cooperation: coercion and persuasion. This boils down to using the carrot and the stick to increase the cooperation of partners. The coercion tactic is about making use of the buyer power and the persuasion tactic means relying on reciprocal investments [54]. For instance in the Truck1 and Truck2 case the willingness and capability to use EDI is a strong supplier selection criterion. The truck companies use their buyer's power because if suppliers do not cooperate then they loose their sales. But at the same time it is reported that both companies also relied on the persuasion tactic. They both were (and still are) actively involved in industry networks that promoted integration in the industry through for instance the establishment of industry data standards. Another example of the persuasion tactic is found in the FMCG case. In this case the portal usage is stimulated with reciprocal investments such as portal related training and support of the supplier by FMCG.

The VAN structure with an external integration service provider was not adopted much among the case companies (only FMCG). But some are thinking about it or have such a structure in development, because (among other reasons) it is expected to enable a higher breadth and volume.

# 6.2.3 Case observations

In Table 15 the current breadth and volume at the studied companies is listed.

Company	Volume	Breadth
	(% document flow)	(% trading partners)
Pharma1	Currently 20 %, when Suppliers are also connected this is to be 50%	3 (10%) of the most important Contract Manufacturers and the 10 % most important suppliers
Pharma2	Estimated 50 % (mainly larger affiliates are integrated)	35% of local affiliates
Packaging1	Most connected plants are larger therefore volume is probably higher then 20 %	12% of the plant locations in the network are connected
Packaging2	Estimated 60 %	Internally 100% of the 7 paper mills and 60% of the 400 corrugated plants. 25% of the external corrugated plants are integrated
Truck1	95 %	90% largest suppliers (rest manual or with web-EDI)
Truck2	95%	80% largest suppliers (10 % with web-EDI, rest manual)
Electronic1	The failed project was aimed at the highest possible volume %. The divisions currently have 5 % EDI and 30 % Supplier Portal volume	The failed project aimed at all suppliers of direct items. The divisions currently have EDI connections with less then 5% of the partners and with 25% supplier portal relations
Electronic2	100% of the internal transactional document streams are supported. Only in exceptions manual intervention is required.	The 40 internal suppliers are all connected. A few external suppliers are also connected (<%5)
Electronic3	The suppliers who are to be integrated account for the most part of the document volume.	20-100 most important suppliers are to be integrated. They are Contract Manufacturers and therefore the supplier base will be reduced (currently 1000)
Electronic4	Peer-to-peer covers 90% of the volume produced by the contract manufacturers Portal 15%	P2P: Top 10 Contract Manufacturers Portal: 15 % of BOM suppliers E-invoice: aims to integrate 70-80 % of all suppliers
FMCG	EDI <5% Portal <10% e-invoicing < 10%	EDI 2 large suppliers with high consumption Portal: +-100 of the larger BOM suppliers and contract manufacturers (total +- 5000 of which 2500 active) e-invoicing: used by +- 250 suppliers of indirect and direct items

#### Table 15: Breadth and Volume

The case observations can be summarized by the following:

- The truck manufacturing companies reach the highest breadth and volume using mature technology because of their longer engagement in Supplier Integration Systems, the collaborative culture in the industry, strong and long-term relationships, the wide diffusion of industry standards, the homogeneity in the industry and strong competitive pressure to produce efficiently.
- Internal integration reaches a higher breadth and volume then external integration, because it is considered easier and because it is considered a pre-condition of external integration
- Supplier Integration Systems breadth and volume levels evolve gradually based on a separate business case for each partner and usually starting with the most important partners first
- Partner cooperation is considered essential to achieve high breadth and volume and is achieved through persuasion and coercion tactics

# 6.3 Process and Information diversity

Diversity is, besides breadth and volume (see section 6.2), an important indicator of Supplier Integration System usage [56]. It explains what the system is used for. Higher diversity means that the systems functionality scope is wider. In general a higher diversity indicates a higher level of integration. First the diversity in general is discussed here. Then the processes and information types are discussed in more detail in the next subsections.

# 6.3.1 Diversity

In many cases it appears that the diversity is different for different partners. With some partners only a view documents are shared and a view processes are supported while with others the diversity is much higher. When looking only at these highly integrated partners the diversity is generally high. This means that, concerning planning and operational purchasing, there are few manual tasks left (only exception handling is often manual). But this does not mean that the maximum possible amount of information is shared. The percentages in Table 16 are relative to the total amount of document types shared. Especially concerning planning, exception handling and confirmations, more information could be shared to increase visibility.

Company	Diversity (% document types concerning planning and operational purchasing)
Pharma1	All document types in use (excluding exceptions) are supported, but not for all partners.
Pharma2	80 % of the document types exchanged is supported (for most partners)
Packaging1	All document types (excluding exceptions) concerning planning and operations are supported for all integrated plants
Packaging2	Most document types are supported, but not with external partners
Truck1	All document types (excluding exceptions) in use are supported, but not for all partners
Truck2	All document types (excluding exceptions) in use are supported, but not for all partners
Electronic1	Most document types are shared with e-faxes, e-mail or manually. The portal only covers some
	concerning ordering, forecasts and confirmations
Electronic2	Mainly transactional documents. Only some of the planning documents with a few partners
Electronic3	All documents types in scope are to be supported.
Electronic4	P2P is most diverse covering all document types in use (excluding exceptions). Portal is less divers,
	and e-invoicing is least diverse.
FMCG	The three systems are limited in diversity. They cover different types.

Table 16: Diversity

#### 6.3.2 Processes

The supported processes at the case companies are listed in Table 17. This table shows what processes are supported for those partners that make full use of the system. In most cases plan exchange, scheduling & ordering, receiving & expediting and invoicing are supported in some way. But when looking at these processes in more detail they appear different and are supported in various ways and degrees. They are not as standard as they seem. Even the most standard process of scheduling & ordering is different in the various cases. In some cases the VMI concept is applied or running forecasts are used as schedules, making the action of sending orders redundant. In some cases explicit order confirmations are required while in others it is agreed in the contracts that every order (within certain boundaries) is delivered, making confirmations implicit. Many of the choices to adopt a certain process are prompted by keeping it as simple as possible. In this way it is easier to automate the processes with the Supplier integration system.

In most of the observed cases during the adoption process the diversity of the system gradually increased up to the current levels. New functionalities to support processes and information types were added. In many of these cases the functionalities were added according their place in the total process from plan to execution and payment. The processes are sequential and that order is kept when processes are introduced. Therefore a general order

of development was that first some plan exchange and ordering was introduced. In later stages of development receiving & expediting and invoicing was introduced in the system. But this general order of development was affected by the problem owner in the focal company. This problem owner, in becoming the integration champion, advocates support for the processes they want integrated first. For instance in the Electronic4 case the e-invoicing system focusing on invoicing only is strongly advocated by the purchasing department, because suppliers complained about payment problems.

Invoicing was in many cases regarded as a special process with its own difficulties and problems. Not only were specific e-invoicing systems setup, but in other cases invoicing was not supported because this was too difficult. This was because the process is not standardized enough across companies and because the regulations for invoicing across different countries also differ considerably. Also it was reported that it requires more trust in the supplier then the other processes, "because it is about the money".

It is striking that large parts of the planning process and the operational purchasing process are not or hardly supported. Local domain planning is not supported by any of the Supplier Integration systems. It is performed in the back end systems, based on information that is locally available. Negotiation & exception handling is hardly supported. Negotiation about plans is mostly not done. Usually the plans are implicitly accepted when they are within the contractual agreements. Exceptional orders are possible within most systems but most exceptions have to be handled manually. Evaluation of plans and compensation schemes are not supported beyond some monitoring functionalities. This is done manually. In the operational purchasing process the verification is not supported at all. In case of wrong or bad products the management of returns and its consequences is performed manually.

		Plann	ing Processes		Operational Processes						
Company	Local Domain Planning	Plan Exchange	Negotiation & Exception	Evaluation	Scheduling & Ordering	Receiving & expediting	Verification	Invoicing & follow up			
Pharma1		Forecasts			V	V		partially			
Pharma2		Mostly VMI	Partners can correct for exceptions		V	V					
Packaging1		Mostly VMI	•	Some monitoring	V	V		V			
Packaging2		Mostly VMI			V	V		Internal only			
Truck1		Forecast and JIT			V	V		V			
Truck2		Forecast and JIT			V	V		V			
Electronic1		Mostly VMI	important suppliers only		V	V					
Electronic2		some			V	V		V			
Electronic3		VMI planned			Planned	Planned		planned			
Electronic4		System- to-system			Portal and system-to- system	Portal and system-to-system		All three systems			
FMCG		EDI and Portal			Portal			e- invoice			

Table 17: Processes supported

### 6.3.3 Information types

The processes described in the last section require the exchange of information. The shared information types are listed in Table 18.

Information is only shared on a need to know basis. The information types of capacity, inventory position, POS data, capacities and lead times are only needed when the VMI principle is applied. In the cases where no order confirmation is send the order is implicitly confirmed according to the contractual agreements. Good receipts do not include the results of the verification process. They only confirm receiving the goods. Usually barcode scanning supports this.

Trust was reported as an important factor affecting the level of information sharing. For instance in the Pharma1 case the VMI principle requiring more detailed information then requirement forecasts was not an option because this would reveal to much about the production process. The suppliers are not trusted enough not to misuse this information.

		Information shared												
Company	Inventory position	Point-of-sale data / sales forecasts	Forecast	Capacity	Lead time	Order	Order Confirmation	Advanced Shipment Notification	Good receipts	Invoice				
Pharma1			V			V	1/2	V	1/2	V				
Pharma2	V	V	V			V	V							
Packaging1	V			V	V	V		V	V	V				
Packaging2	V	V				V		V	V	1/2				
Truck1			V			V		V	V	V				
Truck2			V			V		V	V	V				
Electronic1	V		V			V	V	V						
Electronic2			some			V		V	V	V				
Electronic3	V	V				V		V	V	V				
Electronic4			Peer-			Portal	Portal	Portal	Portal	All 3				
			to-peer			&P2P	&P2P	&P2P	&P2P					
FMCG	EDI	EDI	Portal			Portal	Portal			e-invoice				

**Table 18: Information shared** 

The case observations can be summarized by the following:

- The diversity of information types and supported processes is increased gradually over time.
- The diversity of information types and supported processes varies among partners because of the different capabilities of the partner and the different requirements due to for instance the application of VMI with only a few partners.
- Invoicing is considered a more difficult process to integrate because in practice it varies more among different companies, faces different regulations in different countries and requires more trust in the partner
- Trust was reported as an important factor affecting the level of information sharing

# 6.4 System Intelligence and Integration level

System intelligence describes what kind of decision support the system provides. The integration level with back end systems describes what manual task and delays occur between the moment of a relevant change in the back end data and the delivery of the message to the partner.

*System intelligence* is not widely incorporated in the Supplier Integration Systems. Most of the Supplier Integration Systems can best be seen as a messaging service. They deliver messages and transform them according to the agreed message standard. Logistics intelligence (e.g. decision support) is not supported. This is supported by back end systems. The Supplier Integration Systems do support the tracking and tracing of messages (e.g. message delivery receipts). The systems do monitor their own performance as a "mail carrier". Also in some cases business rules trigger alerts when certain required data is not available or exceeds exceptional values. Higher levels of system intelligence are not expected. Back end systems are regarded to be more suitable for decision support.

The VAN structure with an external integration service provider is expected to provide more opportunities to increase system intelligence in combination with outsourcing services of the third party. But this is not yet put into practice.

Also concerning the *integration level* the case companies do not completely rely on the technology. It also often desired that there are manual checks of the messages by logistics professionals. This is illustrated by the Pharmal respondent: "garbage in is garbage out and therefore messages need to be checked before they are sent". Especially with respect to the forecasts this is considered very important.

# 6.4.1 Case observations

In Table 19 the system intelligence and the back end integration level are summarized.

Company	System Intelligence	Integration level
Pharma1	Only in back end modules. Message delivery receipts and error notifications.	Batch delayed by manual checks before release of any message
Pharma2	There is an advanced planning module integrated. Also: Track and trace capabilities, message delivery receipts and error notifications.	Near real time, batched & delays by manual checks before release concerning planning
Packaging1	Message delivery receipts and simple Business rules only: resulting in error notifications.	Near real time and batch delayed by manual checks before release concerning planning
Packaging2	Minor system intelligent functions in first tier back end. Message delivery receipts and error notifications.	Idem as Packaging1
Truck1	Messaging only. There are delivery receipts and error notifications when delivery fails	Near real time and batch delayed by manual checks for some messages.
Truck2	Idem as Truck1	Idem as Truck1
Electronic1	Only in back end modules. The system only has message delivery receipts and error notifications.	Batch delayed by manual checks before release
Electronic2	The system only has message delivery receipts and error notifications.	Near real time, batched & delays by manual checks before release
Electronic3	The advanced planning capabilities of the logistic service provider are integrated with the sourcing system. This will allow for logistics optimization routines	Near real time, batched & delays by manual checks before release
Electronic4	The peer-to-peer (P2P) system has message delivery receipts and error notifications (exceptional values). The portal uses very strict business rules with most changes requiring special authorization. e-invoicing is purely transactional	Near real time, batched & delays by manual checks before release
FMCG	The system only has message delivery receipts and error notifications.	Batch delayed by manual checks before release

 Table 19: System intelligence & integration level

The case observations can be summarized by the following:

- System intelligence is only applied to support in the form of exception notifications such as failed deliveries of messages or exceptional or missing data, because most of the required system intelligence is provided by back end systems
- Manual checks causing sharing delays and resulting in batch deliveries of messages are required to ensure information quality

# 6.5 Technology

The technological aspects of the structure discussed are the network type, the technology specificity and the process & data standards.

The *network type* used was either a dedicated phone line or the internet. The dedicated phone line is only used for the classical EDI systems in a system-to-system structure. The internet is used in most cases. It is the newer technology and provides more possibilities. The portal structure is only possible because of the internet.

The *technology specificity* says something about the agility of the system. More customized systems are generally more difficult to integrate with new partners and cannot benefit as easily from innovations in the software industry. Off-the-shelve software can benefit from software updates and is generally easier to integrate with new partners. But customization is often required because in some cases the off-the-shelve software does not fit the specific business needs. In all the cases there was a trade-off between agility and level of customization. To some degree customization always was required. The general IT policy and the availability of software at the initial phases of adoption are reported as the two most important factors in the decision. The general IT policy is important because it led to shortcuts in the decision making process. For instance in the Pharma1 case there is a very strong "SAP-unless" policy combined with an outsourced IT function. In others there is still a strong internal IT function combined with a strong legacy of home-made software. These two different legacies influenced the decision making strongly toward either (in the Pharma1 case) off-the-shelve software and in others such as Truck2 towards home made software.

The availability of suitable software is also important because software providers at the time of introduction did not always fit the requirements urging to build home made software or switching to smaller providers. This was the reason FMCG chose a small supplier in spite of their "SAP-unless" policy.

In the VAN structure with an external service provider the software is provided by the service provider.

The *process & data standards* are crucial in the use of any IOS. There are many different standards in use. The case companies prefer to use industry standards since this makes it easier to integrate with partners. But not all industries these standards are fully developed or accepted. Another complicating matter is that companies either operate in different industries or that their suppliers are from different industries. This is very dependent on the industry structure. In the automotive industry the use of standards is widely accepted. There is a strong organization (Odette), with representatives from the entire industry, which promotes integration through the development of these standards. Also in the automotive industry the supplier-buyer relationships are strong and long term. There is a collaborative culture. In the electronics industry the RosettaNet organization provides the de facto standards. In the packaging industry it is Papinet. Also SAP is now beginning to provide its own data standards.

The traditional EDI systems rely on standards based on EDIFACT. This is the older and more mature technology. In many of the newer systems the standards are based on XML. XML is considered cheaper, more transparent, easier to test, more flexible. It was mentioned in the

Electronics3 case: "XML is still developing and EDI is standing still". Successful traditional EDI users (such as the Truck companies) will keep on using the traditional EDI standards. It is expected that this will gradually be complemented with XML standards. But it is not expected to be replaced by it.

The Pharma2 case is a special case. In this industry there were at the time of the initial adoption phases no real industry standards. Therefore they chose to create a special proprietary data standard based on ASCII. The internal IT department had to produce the home made software and was capable to create this proprietary standard. This was easier to implement (low key) then XML based standards. Because of the many different partners and their varying capabilities this was important (not everybody could support XML). ASCII could always be produced and processed (including by low tech partners). But ASCII messages sometimes are more prone to errors when people change the messages.

#### 6.5.1 Case observations

		Technology	
Company	Network type	Technology specificity	Process & Data Standard
Pharma1	Internet	Customized Off-the-shelve software. The organization has a SAP backbone	SAP Idoc (xml) because the entire company is SAP compliant and
Pharma2	Internet	Home made software with off the shelve middleware	Proprietary ASCII based data standard (automatically processed e-mail)
Packaging1	Internet	Off-the-shelve software with custom made components.	Industry standard: Papinet, UIM (XML based)
Packaging2	Internet	Home made and off-the-shelve middleware. Back end systems are very diverse, but mostly SAP	Industry standard: Papinet (XML based)
Truck1	Dedicated phone line	Off-the-shelve software is used. The back en ERP is home made.	EDIFACT from the Odette industry standard
Truck2	Dedicated phone line	Home made software is used	EDIFACT from the Odette industry standard
Electronic1	Internet and Dedicated phone lines	Off-the-shelve software. Very diverse across the company.	Proprietary and some industry standards There is a lack of industry standards, partially due to the diversity of suppliers and the divisions.
Electronic2	Dedicated phone lines	Home made and SAP backbone	Proprietary EDIFACT standard
Electronic3	Internet	The service provider arranges the software. This will be customized.	An XML based RosettaNet industry standard will be used
Electronic4	Internet	There is a SAP backbone. The peer-to- peer integration system is SAP based, the Portal is a customized SAP portal	RosettaNet industry data is used, SAP Idoc standard is used for the Portal, e- invoice will use the third party XML based standard
FMCG	Internet and dedicated phone line	The service provider's own software is used for e-invoicing. Off-the-shelve software from a small provider is used for the Portal, although there is a SAP unless policy this smaller provider was chosen because at the time SAP's ICH was not advanced enough. There is a SAP backbone and SAP middleware is used.	SAP based proprietary standard are used for the Portal, EDI standard is also proprietary, E-invoicing uses an XML based standard of the service provider

The technological aspects of the structure are described in Table 20.

Table 20: Technology

The case observations can be summarized by the following:

• Using standardized and homogeneous software in backend and integration systems would ease integration, but customizations often remain necessary to align with the business.

- The legacy of systems strongly constraints feasible solutions.
- Companies stimulate the diffusion of industry standards because with widely accepted industry standards it becomes easier to integrate with your partners.
- While developments in XML based standards are considered promising, practitioners experience the same difficulties as in using traditional EDI standards

## 6.6 Supply and Relationship Context

The context is an important aspect of the Supplier Integration System structure. It also appears related to other aspects of the structure. The supply context and the relationship context are discussed here. They are listed in Table 21 and Table 22.

#### 6.6.1 Supply Context

The supply context is related to the integration focus. In a plant-to-distribution focus the product types are mainly finished products, while in a plant-to-plant focus the product types are mainly components and other BOM items. Increasingly production is outsourced. Therefore there are many Contract manufacturers and more finished products are in scope. This generally reduces the number of partners. This allows economies of scale but it makes integration also more important to reduce the transaction costs. The outsourcing trend seems to stimulate Supplier Integration Systems as these systems replace the former hierarchy.

"The product types" in scope do seem to affect the adoption process indirectly. They play a role through their characteristics and through their supply market. The supplied product type affects the relationship (see section 6.6.2). Also the volume and the frequency of supplies, which are important criteria in the adoption process (also see section 6.2), depend on the supplied product type and the supply market for these goods. This is what makes suppliers important enough to invest in integration. The supplied product type also determines the commitment of this supplier to the industry of the buyer. In section 6.5 it was shown that a strong industry culture (such as in the automotive industry) is important to stimulate industry cooperation and the establishment of the important industry standards.

The integration of important suppliers is preferred. They are more often integrated and are integrated earlier in the adoption process. Besides delivering more supplies more often, these important suppliers are generally larger and more sophisticated in their IT capabilities. Therefore both benefits are higher and costs are relatively lower.

#### 6.6.2 Relationship context

In most cases the relationship with the partners integrated in the Supplier Integration System are characterized by the respondents as long term and stable. This makes sense because the time to integrate a partner and the costs to do this are generally high. The time to accomplish integration varies between a few weeks in the best circumstances to a year in the worst cases. This depends greatly on the learning curve of the own organization, the diffusion of industry standards and the capabilities of the partner. In most cases the time to accomplish integration is at least a few months, after the contractual agreements have been made.

According to the respondents their relationship with their partners depends on the history together, the cultural match, the amount of relationship specific investments, the level of interdependency, the power distribution between them and the relative importance towards each other.

The commitment and trust in the relationship are important to get the cooperation of suppliers. This is also the reason that there is generally no flexible pricing mechanism

incorporated in the system. Prices are fixed, although rebates are possible when orders are for instance aggregated (larger). But this kind of arrangements are agreed and settled in contracts. Also the contract horizon is generally at least 1 year and often much longer. As was seen in section 6.2, the persuasion tactic in a climate of trust and good relationships is used to achieve cooperation of the partner.

Besides interdependency and trust in the relationship, buyer's power is also considered important in the relationship. The coercion tactic relies on this buying power.

#### 6.6.3 Case observations

The supply context and the relationship context are listed in Table 21 and Table 22.

	Supply context							
Company	Product types in scope	Supplier types						
Pharma1	All direct items are in scope. Order volumes and frequency is the key decision variable	Currently only larger contract manufacturers. Later also important suppliers of direct items. Partners are large companies. Their capabilities are sufficient						
Pharma2	Finished products	Contract manufacturer and internal plant integrate with Partners (which are local affiliates)						
Packaging1	Direct items: the range of items is not very large (10 to 20). They are highly standardized.	The paper mills rely on large economies of scale. Most are internal but some also external.						
Packaging2	Highly standardized direct items	Paper mills rely on large economies of scale and require stability						
Truck1	All truck components	Industry specific of all sizes. Also internal suppliers for most important components						
Truck2	All truck components	Industry specific of all sizes. Also internal suppliers for most important components (e.g. engines)						
Electronic1	Components and materials	Very Diverse: integration only succeeds with some. There are Contract manufacturers, material suppliers and component suppliers						
Electronic2	Finished products and components	Aggregation of plants through internal suppliers.						
Electronic3	Mainly components and finished products	Mainly contract manufacturers with high tech skills.						
Electronic4	Portal and e-invoicing: BOM items include Chemicals, Metals, plastics and components. E-invoicing also includes MRO suppliers	Contract manufacturers, direct suppliers and indirect. Some are very specialized others are more generic						
FMCG	finished products, components and ingredients	Contract manufacturers, direct suppliers and indirect. Some are very specialized others are more generic						

 Table 21: Supply context

	Relationship	context
Company	Relationship type	Relationship asymmetry concerning integration partners
Pharma1	Former colleagues/ long term / stable	The buyer can dictate terms. This buyers power is perceived essential to succeed
Pharma2	The affiliates are relative autonomic, but most are fully owned	Production / central management is the key power.
Packaging1	Very stable and long term, Integration is highly valued	External partners adept to the system. Internally the power is centralized
Packaging2	Very stable and long term, Integration is highly valued	The paper mills are decisive
Truck1	Stable and long term, also in development there is collaboration	Buyer power is relative strong, collaboration in the industry is strong
Truck2	Stable and long term, also in development there is collaboration	Buyer power is relative strong, collaboration in the industry is strong
Electronic1	As diverse as the supplier types. Ranging from arm's length to stable and long term.	Diverse: with some the buyer power is strong while with some the supplier power is strong. (depends market, product and supply characteristics)
Electronic2	Most are internal: therefore power politics are important. Plants do have an incentive to integrate but in higher levels of the organization it is not much supported	The affiliates are relative independent but have little power outside their part of the organization
Electronic3	Long term relationships, Personal contacts. (people know each other), joint product development, mutual investments, relationship specific investments are relative high, VMI (VMI is to be applied with all suppliers in the new situation.)	The interdependency is very high. This enforces a collaborative attitude.
Electronic4	Depending on the market characteristics, with some suppliers and especially sub-contractors there is a very good and long term relationship (Relationship specific investments are high >> switching would be costly). With others there is a more arm's length relationship	Interdependency with Contract Manufacturers is very high (power is divided). For other suppliers the buyer power is relative strong
FMCG	The relationship varies greatly. Some are arm's length and short term and other are much more stable and strategic. The integrated suppliers in general: are in more stable long term relations and are of more strategic importance. Especially contract manufacturers are among them. EDI system partners are the fast runners (high consumption)	Interdependency with Contract Manufacturers is very high (power is divided). For other suppliers the buyer power is relative strong

**Table 22: Relationship context** 

The case observations can be summarized by the following:

- Important suppliers in terms of order volume and frequency are integrated first, because with them the greatest benefits can be achieved.
- Larger partners are usually integrated more successfully because they are generally more sophisticated
- Outsourcing manufacturing reduces the number of suppliers resulting in a smaller but more important supplier base.
- Supplier Integration systems are perceived to be critical to efficiently operate with contract manufacturers and to counter the disintegration caused by outsourcing manufacturing.
- Long term relationships, characterized by trust, and buyers power are considered important to achieve cooperation of suppliers

# 6.7 Adoption process

In Section 2.3.2 the generic phases in the adoption process were described. These are sequentially: awareness, attitude formation, decision making, implementation and confirmation. There and in Chapter 5 it was explained that the adoption process of IOS cannot be seen as a simple straightforward process. In the case interviews it became clear that the Supplier Integration System adoption process indeed can be characterized as a complex, organic, iterative process. They are not finished at once as they grow and evolve over time. This means that the starting point and the ending point of an adoption phase are not clearly recognizable. Organizations are shifting forward and backward between the various adoption phases. They also can be in a different phase at the same time for the various parts of the system's structure. This is best illustrated by the fact that in most cases the systems are not adopted in one big introduction but are gradually developed by starting with only a few partners and few processes and then later by adding more partners and processes. By going through the adoption process iteratively the organization goes through a learning curve. Another example is the technology used as for instance data standards are in constant development.

Some characteristics of each phase could be observed as respondents talked about the history of their system. In the following sub sections these are discussed.

#### 6.7.1 Awareness and attitude formation

The cases made clear that awareness and attitude formation are tacit intangible processes. Through interactions with the environment the organization learns about new innovations and forms an attitude towards them. This happens both informally and formally. A formal way is illustrated by FMCG were a specialized department was created to study the possibilities of ebusiness. Partners, consultants, software vendors, new employees, academics and the media are reported to influence this process. It is clear that this is not straightforward and occurs dynamically. It is also only at best partially a rational process.

This is most frankly illustrated in the Packaging1 case. They started with integration systems during the highs of the internet-hype because of partner pressure. But it later became clear that at the time both the partners and the internal organization were not capable to integrate successfully. A long learning curve and some crucial changes were required before the system could become a success. This was one of the reasons that they started to use their Supplier Integration System for internal integration.

In the Electronic3 case the awareness and attitude formation was highly influenced by a newly hired senior manager who became the champion of a new Supplier Integration System. Coming from the military this person brought in much experience in logistics and started to advocate his view on logistics. He quickly found key supporters but also found large resistance to change.

Another example is the Truck1 case. New developments in logistics and EDI triggered the truck company to seek out EDI. Large government grants helped to push decision making in favor of the then new innovation of traditional EDI systems. Without these grants that stimulated industry wide collaboration, implementation would have been much more difficult.

#### 6.7.2 Decision: costs, risks and benefits

The decision making phase comprises of both high level "go or no-go" decisions and decisions about detailed aspects of the structure. These decisions are based on the attitude that was formed before. The decision maker's attitude (or perception) towards a Supplier Integration System is (or should be) formed by the best possible estimation of the costs, risks and benefits of the system and its various aspects. In Table 23 and Table 24 the perceived and/or estimated costs, risks and benefits are listed for the various case companies. As was shown in section 6.7.1 the attitude formation was also influenced by other factors. It also

became clear during the interviews that both costs and benefits are difficult to quantify and specify. Decision making was therefore not always completely rationalized.

Perceived benefits of the Supplier Integration System	Pharma1	Pharma2	Packaging1	Packagin2	TruckI	Truck2	Electronic1	Electronic2	Electronic3	Electronic4	FMCG
More information can be handled				V			V				
Better quality information (more detail and better aggregated)	V			V						V	
Increased reliability of information	V			V						V	V
Faster communication	V	V		V	V	V	V			V	V
Reduction of lead times	V	V						V	V		
Reduction of errors	V	V	V		V	V	V			V	V
Standardization and alignment as a by product	V										
Reduced stock level	V	V	V	V	V	V	V	V	V		
Increased efficiency for partners >> leading to better prices	V										
Faster reactions on contingencies	V						V				
Increased visibility		V	V				V	V	V	V	V
Planning and forecasting abilities increase (advanced planning)		V		V							
Increased administrative and transactional efficiency		V	V	V	V	V	V			V	V
Cycle time reduction	V	V						V	V		
Increased deliver reliability			V					V	V		
Stock flexibility			V								
Product flexibility increased				V							
JIT concept becomes possible				V	V	V					
lower overhead costs									V		
Reduction of managerial complexity (smaller supplier base)									V		
Cost reduction finance department											V
Payments are faster										V	V
Table 23. Benefits		·								l	

<b>I</b> able	23:	Benefits	

Phama1	Pharma2	Packaging1	Packagin2	TruckI	Truck2	Electronic1	Electronic2	Electronic3	Electronic4	FMCG
V										
V			V			V				
V	V	V				V	V		V	
V	V	V		V	V	V	V		V	
								V	V	
		V	V	V	V	V	V	V	V	V
			V							
			V							
				V	V					
			V							
				V	V					
								V		
									V	V
									V	
	V           V           V           V           V	V           V           V           V           V	V         V           V         V           V         V           V         V           V         V	V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V           V         V	V     V       V     V	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$

Table 24: Costs and risks

#### 6.7.3 Implementation

The implementation of a Supplier Integration System was in the case companies a prudent process. Usually a pilot phase started with only a few partners and processes. Partners and process diversity was later added gradually. A learning curve could be observed as the time to integrate each additional partner shortened. Formal evaluations are not reported. They

constantly occur informally though, as new attitudes towards aspects of the structure are constantly formed. Companies gain understanding about Supplier Integration Systems while they are engaged with them.

Prudence in the adoption process does not appear superfluous as most successful cases had a prudent approach while the large failed project at Electronics1 is an example of wanting to do too much at once.

The implementation of a Supplier Integration System has many similarities with large enterprise system implementations. Many processes, tasks and responsibilities must be changed to optimally benefit from the new system. Like in enterprise system implementations the organization undergoes a transformation. Resistance to change must be overcome. For instance in some cases the organizational culture is an impeding factor. In the Electronic2 case; the dominant culture is oriented towards technical product innovations and not orientated towards SCM. This is illustrated by the respondent: "New gadgets and technological innovations are cool, SCM is not".

Also the cooperation of partners is important during adoption. It seems that most success is achieved in the cases with stable and long term relationships characterized by a climate of trust. "The partner must be convinced that the change is beneficial" according to the respondent of FMCG.

#### 6.7.4 Case summary on adoption

The case observations can be summarized by the following:

- Organizations are prudent in the adoption of Supplier Integration Systems
- Decision making is not completely rationalized as costs, risks and benefits are difficult to specify and quantify.
- The adoption of Supplier Integration Systems is an intangible and iterative process

#### 6.8 Case conclusions

This chapter has treated the structure of Supplier Integration Systems and the relation with its adoption process from a practice oriented point of view, thus answering sub-questions 1d, 3b and 3c. The structure (answering 1d) has been treated extensively in this chapter. At same place in this chapter the explanation the respondents could give (answering 3c) for an aspect of their structure has been treated. The adoption process (answering 3b) has been treated in Section 6.7.

Although all case companies have adopted some sort of Supplier Integration System, the structures they adopted differ considerably. Three basic structures are observed and they are used in complementary ways. A Supplier Integration System can also be part of a larger supply chain platform offering more functions and features, also aimed at servicing the downstream supply chain.

The maturity of the case companies concerning Supplier Integration Systems varies greatly. It has become clear that among the studied cases the truck companies are most mature. But compared to what is theoretically possible the gap is even larger. The truck companies rely on traditional EDI which is the most mature technology. The case companies are not as mature as could have been. After all the internet hyped 10 years ago and classical EDI is more then two decades old. Many of the case companies do not have systems with a large breadth, volume or diversity, while the respondents say that they do pursue them. System intelligence, system flexibility and the integration level also could be much higher. But these levels seem to be more according to their current preferences.

Integration is clearly more difficult then the concept at first seems to indicate. Implementation costs are high and benefits are difficult to quantify. Concerning the technology and the legacy of integration and backend systems the path dependency is high. According to the respondents

this is caused by high switching costs and a strongly felt constraint of back-end and legacy systems.

Heterogeneity of partners and internal divisions is considered a large problem. The case companies operating in a single cohesive industry, having many dedicated partners and having a collaborative culture seem to integrate easier. Also in the cases a high level of integration internally is considered to be a prerequisite before one can integrate externally. But at the same time integration systems can help to integrate internally and help to enforce standardization. It appeared that the same systems are used for internal supplier as well as external supplier integration. The difference between EAI and B2B is small according to the respondents.

Most of the case observations are about the structure and the explanation the respondents could give for their situation. The adoption process itself was difficult to describe. But they could be characterized as a complex, organic, iterative process. They are not finished at once as they grow and evolve over time. This means that the starting point and the ending point of an adoption phase are not clearly recognizable. Organizations are shifting forward and backward between the various adoption phases. They also can be in a different phase at the same time for the various parts of the system's structure.

In the next chapter the findings presented in this chapter combined with findings from the literature study are discussed more comprehensively. Here in Table 25 and Table 26 the case conclusions from this chapter are listed. It must be noted that these are only the conclusions of this particular set of cases based on the answers provided by the respondents. They can not be instantly generalized to other cases.

	The system-to-system structure is most difficult to implement but enables a high degree of integration
Basic	
structures	The VAN structure can be combined with outsourcing services by an integration service provider
structures	but makes one dependent on a sometimes costly and not always competent service provider.
	The supplier portal structure is the most accessible structure but only provides partial integration, thus still requiring manual tasks.
	The three basic structures have distinct governance structures and are used in a complimentary way for both internal and external integration.
	The truck manufacturing companies reach the highest breadth and volume using mature technology
D	because of their longer engagement in Supplier Integration Systems, the collaborative culture in the
Breadth & volume	industry, strong and long-term relationships, the wide diffusion of industry standards, the homogeneity in the industry and strong competitive pressure to produce efficiently.
volume	Internal integration reaches a higher breadth and volume then external integration, because it is
	considered easier and because it is considered a pre-condition of external integration Supplier Integration Systems breadth and volume levels evolve gradually based on a separate
	business case for each partner and usually starting with the most important partners first
	Partner cooperation is considered essential to achieve high breadth and volume and is achieved
	through persuasion and coercion tactics
	The diversity of information types and supported processes is increased gradually over time.
Diversity	The diversity of information types and supported processes varies among partners because of the different consultance due to for instance the
Diversity	different capabilities of the partner and the different requirements due to for instance the application of VMI with only a few partners.
	Invoicing is considered a more difficult process to integrate because in practice it varies more among different companies, faces different regulations in different countries and requires more trust
	in the partner
	Trust was reported as an important factor affecting the level of information sharing
System	System intelligence is only applied to support in the form of exception notifications such as failed
Intelligence	deliveries of messages or exceptional or missing data, because most of the required system
&	intelligence is provided by back end systems
a Integration	Manual checks causing sharing delays and resulting in batch deliveries of messages are required to
Delays	ensure information quality
Table 25. C	1 2

Table 25: Summarized case observations (part 1)

	Using standardized and homogeneous software in backend and integration systems would ease
	integration, but customizations often remain necessary to align with the business.
Technology	The legacy of systems strongly constraints feasible solutions.
	Companies stimulate the diffusion of industry standards because with widely accepted industry
	standards it becomes easier to integrate with your partners.
	While developments in XML based standards are considered promising, practitioners experience
	the same difficulties as in using traditional EDI standards
	Important suppliers in terms of order volume and frequency are integrated first, because with them
	the greatest benefits can be achieved.
	Larger partners are usually integrated more successfully because they are generally more
Relationship	sophisticated
Context	Outsourcing manufacturing reduces the number of suppliers resulting in a smaller but more
	important supplier base.
	Supplier Integration systems are perceived to be critical to efficiently operate with contract
	manufacturers and to counter the disintegration caused by outsourcing manufacturing.
	Long term relationships, characterized by trust, and buyers power are considered important to
	achieve cooperation of suppliers
	Organizations are prudent in the adoption of Supplier Integration Systems
Adoption	Decision making is not completely rationalized as costs, risks and benefits are difficult to specify
Process	and quantify.
	The adoption of Supplier Integration Systems is an intangible and iterative process

Table 26: Summarized case conclusions (part 2)

# 7 Conclusion and Recommendations

In the last three chapters the sub-questions posed in chapter 3 have been answered. In this chapter these answers are used to answer this study's main research question.

# What is the impact of Supplier Integration System adoption processes on Supplier Integration System structures of manufacturing companies?

By answering this question our understanding of Supplier Integration System structures and where they come from, can be improved. This chapter will give some answers to this question in two ways. First in section 7.1 the conclusions of this study will boil down to a refinement of the adoption process model, as was suggested in Section 3.1.2. This refined model should give more insights in the adoption process and its impact on the structure. Naturally this is constraint by the limitations of the study. Then this chapter finishes with the recommendations for practitioners and future research projects.

# 7.1 Conclusions

This section deals with the structure and the adoption process of Supplier Integration Systems and the relation between the two. The conclusions on the structure are first explained in Section 7.1.1. Then in Section 7.1.2 the conclusions on the adoption process are presented. This includes refinements to the second order factor model of adoption. The conclusions on the impact of this adoption process are discussed in Section 7.1.3. Finally the limitations of these conclusions are discussed in Section 7.1.4.

# 7.1.1 Structure

This study demonstrates that Supplier Integration Systems do not have a uniform structure as this sample of eleven cases contains various structures. Three basic structures could be observed and these basic structures could be complemented with a range of characteristics to distinguish individual systems. Systems are used to integrate internal supplier as well as external suppliers

The structures of the studied companies can be considered dynamic. While the basic aspects of the structure remain the same during its entire life cycle, many aspects change. The cases suggest that Supplier Integration Systems usually start small and usually end up with more volume, diversity and breadth. Some case systems are also expected to evolve more in this respect. Also in the cases the technologies in use are updated during the systems life cycle. The evolving nature of the Supplier Integration System structures observed in this study should be confirmed by more studies before this can be generalized though.

It is noticeable that the Supplier Integration System structures in the studied cases are less advanced and less comprehensive then could be expected when looking at theoretical possibilities and the reputation of the companies. The cases used in the study are selected because these companies are the usual suspects to have sophisticated integration systems. They are very large multinationals with much resources and a strong presence in their markets. They are leading companies and could benefit much from high integration levels. But in terms of breadth, volume and diversity most case companies are not mature. Especially invoicing has a very low support rate. And many of the systems focus on the internal suppliers because this is thought to be easier as internal partner cooperation can be enforced by top management. Also system intelligence is limited to monitoring of messages and some notifications of errors and exceptional values. System intelligence in terms of for instance decision support or smart agents supporting planning negotiations and exception handling seams a far fetch. The integration level is usually not real time as information is often sent in batch messages and delayed due to manual checks. The systems are rigid as changing partners and changing processes, usually takes months and in the best cases weeks. Dynamic pricing mechanisms or flexible sourcing mechanisms were considered not possible.

It can be concluded that there is a variety of Supplier Integration System structures, which are used by companies in a complimentary way. Also it can be concluded that these structures can have an evolving nature and that there are still companies having less advanced structures then could be expected.

#### 7.1.2 Adoption process

The current system structure and its evolution are determined by the adoption process. In this study the adoption model of Kurnia and Johnston [24] was used (also see Chapter 5). The results justify an extension of their model, but they also indicate that most elements can be confirmed.

**The process logical aspect** of adoption is the most evident element that could be confirmed. The case studies showed that it takes time to arrive at their specific structure. All the cases were projects taking years. They were undergoing significant changes to their structure during their adoption. During this process the organizational actions are influenced by many kinds of factors and at the same time the organization is changing itself and its environment. Also over a longer period of time it becomes clear that the significance and nature of factors change over time. They can only be seen as conditions that may lead to adoption.

The emergent aspect of the adoption process is also confirmed. The case companies are not in total control of their environment, but are not helpless victims either. Deliberate actions and strategies to influence the environment and partners, to improve the organizational capabilities and to adapt the nature of the technology are successfully applied in practice. Prime examples are the active involvement in industry associations to develop industry standards and the tactics to achieve partner cooperation. But also opportunistic as well as helpless behavior occurs. Companies seize opportunistically the right moment to implement their structure. Take for instance the case example where SAP developed a new integration module resulting in the implementation of this integration model at some of the case companies and the Truck company seizing government grants to help finance their integration system. But companies also blindly follow hypes and partner pressure, as the respondent of the Packaging1 company dared to admit.

**The firm/inter-firm aspect** of the Supplier Integration System adoption process and structure also was confirmed. The inter-firm level appeared to be an important part of the structure. In this plane the system is situated and success factors such as partner cooperation are influenced by the firm with persuasion and coercion tactics. Therefore the choice to distinguish the categories "Environment", which cannot be influenced by the organization, and "Supply Chain / Industry Structures", which can be influenced by the organization, is supported. This does not mean that the intra-firm level of analysis is not important. In some of the studied companies the internal supply chain is the primary integration focus.

The results of this study also justify an extension of the model. In Figure 15 the second order adoption model by Kurnia and Johnston [24] is adapted to fit the results of this study (compare Figure 10). Four changes in the model are proposed.

**1.** The case studies show that the assumption of a dichotomous outcome of the adoption process is an invalid simplification of reality. As was explained in section 7.1.1 the case studies show varying and dynamic structures. Companies do not arrive at their ultimate

structure at once. It takes time to arrive at a certain structure and during further iterations of the adoption process the structure keeps changing. In Figure 15 this is represented by the box tagged "a Supplier Integration System structure".

**2.** This study showed that an implemented Supplier Integration System structure can impact the organization and its environment and may lead to new innovations in the technologies used. A possible explanation, confirmed by the respondents, is that organizations and partners go through a learning curve as they gain experience with using Supplier Integration Systems. As they and their partners gain experience they can and will improve for instance the industry standards they are using. Also other influences on the relationship are noticed in some of the cases. Because of the mutual relationship specific investments involved in the Supplier Integration System the relationship was strengthened and partner trust was increased. In Figure 15 this is represented by the extra arrow pointing back from the box tagged "a Supplier Integration System structure".

**3.** The results show an evolving and iterative adoption process. It is almost impossible to pin point the end of the adoption process, because during the systems life cycle changes to the structure remain occurring. Therefore the adoption process can be considered a cyclic and iterative process. In Figure 14 the adoption process is schematically depicted. While the original model [13], already discussed in Section 2.3.2., also is explained having an iterative nature, the representation is linear. To emphasize the cyclic and iterative nature of adoption, the figure is a circle. This figure can be illustrated by the example of the Truck cases. Their current data standards are not the same as they started with. They had implemented their EDI-system, but at a later point in time they became aware of new developments concerning the industry standard. They formed a positive attitude and decided to implement it with key suppliers. It turned out to be a positive change and more partners were beginning to use it.



Figure 14: Adoption Process (with adaptations [13])

The circle in Figure 14 is used in Figure 15. The organization's actions are represented by the circle in the middle. The circle is the smaller version of Figure 14 with the five phases of the adoption process. It is meant to represent the dynamics and iterative nature of the adoption process. Companies are at different levels of analysis involved in different phases of the adoption process. In every phase the factors impact organization's actions and the organization's actions impact these factors. These actions may lead to the adoption of a specific Supplier Integration System structure. But from the adoption and use of this system the organization and its partners learn through the formal and informal evaluation of the system and the actions performed. These organizational learning actions impact the factors. Therefore there is an arrow added that points from the "Supplier Integration System structure" back to the circle.

**4.** The factors are adapted to the specific situation of Supplier Integration System adoption. These conditions affecting the organization's actions are listed in the boxes in Figure 15. They are grouped according to technology, organization, environment and industry structure like in the original model. These categories represent a wide number of factors, which are only summarized by the lists here. While the precise factors are not exactly the same as in the original model they confirm the original argument that factors must be seen as conditions that may lead to adoption. Therefore this model does not presume to present a complete list, nor does it mean to suggest that all the factors are always important. Also each factor can manifest itself in different ways and many factors are related and/or impact each other.

**External factors** are the only factors, like in the original model, the organization has no real direct influence in. But these factors do impact the organization's actions. Government regulations are important because they set constraints. The government may also provide support in the form of grants or knowledge like in the Truck1 case. A competitive and global marketplace and low efficiencies forces companies to change like in the Electronic4 case. The current outsourcing trend and the integration efforts are both directions of this change. Outsourcing, as illustrated in the Electronic3 case, is also an example of a socio-economic trend that both forces and eases integration.

The perceived **nature of the technology** is a set of conditions the organization can affect to a certain extend but also strongly impacts the organization's actions. The integration benefits are the perceived benefits of integration (see Table 23 in Chapter 6) and are the prime reason to adopt a Supplier Integration System. The compatibility with the business objectives factor reflects the strategic relevance of integration. Execution flexibility (the effort required to integrate another partner) is an important aspect because it is required to achieve high breadth and volume and because it decreases the partner switching costs. This in turn depends on the basic structure, organizational capabilities (especially integration experience) and the diffusion and use of industry standards. Developing flexibility is the degree and costs to which new functionality can be added. This is important to achieve diversity (see Section 6.3). Like benefits the costs and the perceived risks are important in any business case. The costs to implement the new system and to operate the system are important but difficult to specify (see Section 6.7.2).

The conditions in this category can be affected by dedicated teams exploring e-business opportunities, external expertise, research and development.

The **capability of the organization** is a set of conditions the organization can affect strongly but also impacts the organization's actions. Top management commitment, clear strategic vision, communication openness and organizational flexibility are critical for the organizational transformation. This could be compared with any major enterprise system implementation. The required skills of the staff and partner staff are enhanced by adequate training and education capabilities. Performance measures are required for the evaluation the system. To adopt a Supplier Integration System it is a pre-condition to have adequate IT infrastructure internally. This is also the reason that integration of the internal supply chain is considered a pre-condition before external integration. Integration experience represents the learning curve of the organization. As companies gain experience their capabilities increase. The conditions in this category can be affected through the development of vision and support by top management, hiring external expertise and change management.

The **Supply Chain / Industry Structures** represents a set of conditions which are external to the organization but still can be affected. This category strongly affects the adoption process as was shown in Section 6.6. The relationship in terms of power and trust is important to achieve cooperation and enhance the breadth, volume and diversity of the system (see Section 6.2 and 6.3). Also when partners can benefit they are more likely to cooperate. The

capabilities of the partner organization determine the costs to integrate. The diffusion and use of industry standards (see Section 6.5) is a very important factor because through network effects the value of using this standard would increase when more partner use the standard. It also increases execution flexibility.

The conditions in this category can be affected by active involvement in industry association and relationship specific investments. The implementation of a Supplier Integration System itself was also affecting this category through its relationship specific investments, organizational learning by partners and possible vendor lock in.



Figure 15: Extended adoption process model

#### 7.1.3 Adoption process impacts on structure

It is not possible to tell exactly what *the* impact of *the* adoption process is on *the* structure. A conclusion from this study is that there is not one single path to arrive at a certain Supplier Integration System structure. The case companies arrived at similar basic structures in different ways. Some companies took more deliberate actions to exert their influence, while others behaved more opportunistically. But also the question what the optimal structure is in a given situation, taking into account the possibilities to positively affect this situation and the costs to achieve this structure, could not be answered conclusively. In general the respondents consider higher breadth, volume, diversity, and flexibility positive. Although benefits are difficult to quantify and costs are difficult to specify, higher breadth, volume and diversity are thought to increase the benefits from integration at decreasing marginal costs. Flexibility is considered important because the structure must be adaptable to changing circumstances (e.g. a changing market).

The impacts of the adoption process comprise of the deliberate actions of the organizations and comprise of the opportunistic reactions to the conditions (the factors in Figure 15). While no conclusive answer can be given on the complete impact of the adoption process on the structure, the cases indicate some impacts, concerning specific aspects of the structure. The following explanations could be synthesized from the case studies:

The **basic structure** of the studied systems is mainly determined by the perceived nature of the technology. The portal-structure is easy to implement and is a very flexible technology when compared to the other structures. But the relative advantage of the other two basic structures is the higher degree of integration resulting in higher operating efficiencies.

The **breadth and volume** of the studied systems is mainly determined by the cooperation of the partner and the costs to connect a partner. Cooperation can be stimulated through persuasion and coercion tactics. This is mediated by the nature of the relationship. The relationship can be characterized by the power distribution and level of trust. Coercion tactics work best when there is much buyer's power. And persuasion tactics work best when the relationship is based on high levels of trust. The costs depend largely on the skill and infrastructure of all the involved parties, the existence and use of an industry standard, the execution flexibility, the basic structure type and the level of cooperation. Industry standards can actively be promoted in industry associations. Furthermore the cases show that achieving high breadth and volume takes time.

The **diversity** of the studied systems is mainly determined by the expected costs and benefits of adding a process and/or information type and the nature of the relationship. Costs and benefits are contingent on the specific situation. Costs depend largely on the skill and infrastructure of all the involved parties, the existence and use of an industry standard and the level of cooperation. Industry standards can actively be promoted in industry associations. Cooperation can be achieved through persuasion and coercion tactics. In the cases trust was an important factor positively affecting diversity (see Section 6.3)

The **flexibility** of the studied systems is mainly determined by the diffusion and use of industry standards, the organizational capabilities of all the involved parties and the basic structure type. Experience with the system in terms of a positive learning curve was reported to enhance flexibility.

The **integration partner selection** in the studied cases is mainly determined by the costs and benefits of integration with the specific partner. Benefits are dependent on the order volume, frequency and to some extend value of the items (because stock reduction of valuable items reduces inventory costs more). Costs consist of the implementation costs and the operating costs. The costs depend largely on the skill and infrastructure of all the parties, the existence and use of an industry standard, the execution flexibility, the basic structure type and the level of cooperation. Internal supply chain partners are generally preferred because this is considered easier and a pre-condition.

The **technology** of the studied systems is mainly determined by the perceived benefits, the legacy (compatibility) and the future expectations of the technology. The standards used depended largely on the diffusion in the industry as the perceived value of the standard increases with its acceptance among trading partners. The studied companies prefer an industry standard and actively promote this in industry associations. Path dependency and network effects are confirmed by the cases.

#### 7.1.4 Limitations

Some limitations of this explorative research project are the equivocal results and the lack of conclusions that can be generalized for every large company. This is in part caused by the nature of IOS adoption as a complex process and in part by the explorative and qualitative nature of the research approach. The small number of cases does not qualify for a statistically significant conclusion. The reader should therefore not blindly jump to conclusions and

remain prudent. That being said the reader can use this model as its use is illustrated in this thesis. Here some general comments on the applicability are provided.

The refined model is based on a more generic model of inter-organizational systems. But the study focused on large manufacturers and their suppliers and on the integration of operational processes. The integration focus might not matter much to the applicability of the model in other supply chain situations. More downstream or upstream the supply chain the conditions likely have a similar nature. But integration systems that concern the integration of tactical processes are likely to face different conditions.

Furthermore the research method also caused some limitations that should be heeded. The selection of the cases and respondents limits this research to some degree. The studied companies are all very large multinationals. Small and medium sized companies could have a different adoption process as it is likely that they are less able to influence their environment. Most of the case companies have a European background and all the respondents are Dutch. This could have caused some bias to the European/Dutch perspective. This is countered by the fact that all the companies are global. Finally the respondents are professionals with responsibilities overlapping the subject. This made them subject matter experts but could also

Finally it should be noted that the case studies could not be optimally examined. In only four of the eleven cases more then one respondent was interviewed. This could have caused a bias to the soul respondent's perceptions, although this bias could not be found in the data. The respondents were subject matter experts within their organization. Therefore it can be relative safely assumed that their statements about the current status and facts correspond with reality. But answers with causal statements are probably more biased by the perception of the respondent. Also due to the semi-structured nature of the interviews the bias of the researcher could have influenced the respondent's answers. Especially concerning factors and results this could have caused a bias in the answers.

# 7.2 Recommendations

have biased their perception too positively.

The contribution of the study ultimately converges in the recommendations to others. The thesis targets an audience of practitioners and academics. In this section the practitioners are helped with some recommendations about how to apply the model and the lessons learned from the case studies. Then the recommendations to academics are discussed in terms of future research.

#### 7.2.1 Practitioners

The refined model of adoption should help practitioners to better understand their own Supplier Integration System adoption process. This section explains how practitioners can use the results of this study. First the considerations with respect to what one would want to achieve are discussed. This is followed by a reflection on the organization's actions to achieve the desired structure.

The question for companies, which fit in the case study profile, should not only be whether or not to adopt a Supplier Integration System but rather it should also question what structure would be optimal. The Supplier Integration System structure one would want depends on the situation, taking into account the possibilities to positively affect this situation and the costs to achieve this structure. In general higher breadth, volume, diversity, and flexibility are considered positive, although benefits are difficult to quantify and costs are difficult to specify. The most successful cases in this study, which are the truck companies, achieved a breadth and volume of over 80 % combined with high diversity and relative high flexibility in

their industry. Ultimately this would probably be a good objective for any company that fits in the case study profile, as the truck companies can be considered best practices.

To achieve this best practice one can use the refined adoption model presented in this study. But one should approach the adoption as a complex and long term project and take small steps forward. The following steps should be followed through:

- 1. Use the adoption model as a checklist to analyze the situation.
- 2. Identify the conditions, which are not met, and develop deliberate actions to meet them.
- 3. Weigh the costs and benefits of these deliberate actions and alternative opportunistic behavior (seizing the right conditions at the right moment).
- 4. Based on this analysis, choose changes in the structure(s), taking into account that different structures are also successfully used in a complementary way.
- 5. Repeat the previous four steps

When following these steps some recommendations, which would hold in most cases, could be provided:

- 1. Choose the Portal-structure when you prefer low implementation costs and lower risks over lower operating costs or when the capabilities of most partners are at a low level.
- 2. Choose the System-to-system structure when you prefer high integration benefits over higher implementation costs or when there is a widely diffused industry standard.
- 3. Choose the VAN-structure when you want to benefit from integration services, high integration benefits and higher execution flexibility against extra operating costs.
- 4. Choose important suppliers (in terms of order volume, frequency and value) to integrate first.

#### 7.2.2 Future research

Based on the results from this study future research can contribute by studying tactics and their effectiveness, by studying the structure in relation to firm performance, by studying the impact of the Supplier Integration System on the Supply Chain structure and by studying how to achieve flexibility. Future research can also contribute by improving the adoption model.

In this study several tactics were encountered to positively affect the situation. The coercion and persuasion tactics to achieve partner cooperation, the use of innovation teams to study new developments, the active participation in standards associations are prime candidates to be studied for their effectiveness.

It became clear that it is difficult to quantify and specify both costs and benefits of a Supplier Integration System structure (see section 6.7.2). What would be the best structure depends on many factors. It would therefore be interesting to sort out the impact of a structure as a whole. Some crucial aspects such as breadth, integration level and flexibility could then be specifically studied to find their impact on firm performance.

When the Supplier Integration System is placed in the inter-firm environment it is likely to have an impact on the Supply Chain structure. The relationship with partners could be improved and there could be a shift in power or trust. Also the economic relation could be affected as switching costs could increase. Therefore the impact of the Supplier Integration System on the Supply Chain structure would be an interesting subject to study.

It became strikingly clear that it still takes a lot of time to integrate with partners or to change functionality, while flexibility is important. How to improve this flexibility is a technical and organizational problem.

Future research can improve the adoption process model by studying the adoption phases separately. Some conditions could be more important in earlier adoption phases then others while other factors are more important in the implementation phase.

# 8 References

- 1. Palermo, L. *STMicroelectronics Foundries Integration*. 2005 [cited 25-2-2008]; Available from: <u>http://www.edifice.org/Documents/WHITE%20PAPER%20FOUNDRIES%20INTE</u> <u>GRATION%20EXTERNAL.pdf</u>.
- 2. Monczka, R., R. Trent, and R. Handfield, *Purchasing and Supply Chain Management*. 3rd ed. 2005, Mason, Ohio: Thomson South Western.
- 3. Van Weele, A.J., *Purchasing and Supply Chain Management: Analysis, Planning and Practice.* 3rd ed. 2002, London: Thomson Learning.
- 4. Quinn, J.B., Intelligent Enterprise. 1992, New York: Free Press.
- 5. Kraljic, P., *Purchasing must become Supply Management*. Harvard Business Review, 1983 (September-October): p. 109-117.
- 6. Stadtler, H., *Supply Chain Management An Overview*, in *Supply Chain Management and Advanced Planning*, H. Stadtler and C. Kilger, Editors. 2005, Springer: Berlin.
- 7. Archer, N. and Y. Yuan, *Managing business-to-business relationships throughout the e-commerce procurement life cycle*. Internet Research: Electronic Networking Applications and Policy, 2000. **10**(5): p. 385-395.
- 8. Papazoglou, M.P. and P.M.A. Ribbers, *e-Business: Organizational and Technical Foundations*. 2006, Chichester: John Wiley & Sons, Ltd.
- 9. Malhotra, A., S. Gosain, and O.A. El Sawy, *Absorptive capacity configurations in supply chains: Gearing for partner-enabled market knowledge creation*. MIS Quarterly: Management Information Systems, 2005. **29**(1): p. 145-187.
- 10. Chen, F., et al., *Quantifying the Bullwhip Effect in a Simple Supply Chain.* Management Science, 2000. **46**(3): p. pp. 436–443.
- 11. Kaufman, F., *Data systems that cross company boundaries*. Harvard Business Review, 1966. (Jan-Feb).
- 12. Harink, J.H.A., *Internet-technologie in de inkoop ...de kinderschoenen ontgroeid...* in *School of Management and Governance*. 2003, University of Twente: Enschede.
- 13. Narayanan, V.K., *Managing Technology and Innovation for Competitive Advantage*. 2001, New Jersey: Prentice-Hall, Inc.
- 14. Ng, J.K.C. and W.H. Ip, *Web-ERP: the new generation of enterprise resources planning*. Journal of Materials Processing Technology, 2003. **138**: p. 590-593.
- 15. Kaplan, S. and M. Sawhney, *E-Hubs: The New B2B Marketplaces*. Harvard Business Review, 2000(May-June).
- 16. Howard, M., R. Vidgen, and P. Powell, *Automotive e-hubs: Exploring motivations* and barriers to collaboration and interaction. Journal of Strategic Information Systems, 2006. **15**(1): p. 51-75.

- 17. Barratt, M. and K. Rosdahl, *Exploring business-to-business marketsites*. European Journal of Purchasing and Supply Management, 2002. **8**(2): p. 111-122.
- 18. Baglieri, E., R. Secchi, and S. Croom, *Exploring the impact of a supplier portal on the buyer-supplier relationship. The case of Ferrari Auto.* Industrial Marketing Management, 2007. **36**(7 SPEC. ISS.): p. 1010-1017.
- 19. Li, M., et al., *A collaborative application portal for the mould industry*. International Journal of Production Economics, 2005. **96**(2): p. 233-247.
- Shevchenko, A.A. and O.O. Shevchenko, B2B e-hubs in emerging landscape of knowledge based economy. Electronic Commerce Research and Applications, 2005. 4(2): p. 113-123.
- 21. Zeng, A.Z. and B.K. Pathak, Achieving information integration in supply chain management through B2B e-hubs: Concepts and analyses. Industrial Management and Data Systems, 2003. 103(8-9): p. 657-665.
- Dai, Q. and R.J. Kauffman, *To be or not to B2B: Evaluating managerial choices for e-procurement channel adoption*. Information Technology and Management, 2006. 7(2): p. 109-130.
- 23. Supply-Chain-Connect. [cited September 2007]; Available from: http://www.supplychainconnect.com/private\_supply\_hubs.html.
- 24. Kurnia, S. and R.B. Johnston, *The need for a processual view of inter-organizational systems adoption*. Journal of Strategic Information Systems, 2000. **9**(4): p. 295-319.
- 25. Henriksen, H.Z., *Motivators for IOS adoption in Denmark*. Journal of Electronic Commerce in Organizations, 2006. **4**(2): p. 25-38.
- 26. Zhu, K., K.L. Kraemer, and S.X. Xu, *Electronic business adoption by European firms: a cross-country assessment of the facilitators and inhibitors*. European Journal of Information Systems, 2003. **12**: p. 251-268.
- 27. Van Hillegersberg, J., Information Systems for a Networked World, in School of Management and Governance. 2006, University of Twente: Enschede.
- 28. Iacovou, C.L., I. Benbasat, and A.S. Dexter, *Electronic Data Interchange and Small Organizations: Adoption and Impact of Technology*. MIS Quarterly, 1995. **19**(4): p. 465-485.
- 29. Kurnia, S. and R.B. Johnston, *Adoption of efficient consumer response: The issue of mutuality*. Supply Chain Management, 2001. **6**(5): p. 230-241.
- 30. Hart, P. and C. Saunders, *Power and Trust: Critical Factors in the Adoption and Use of Electronic Data Interchange*. Organization Science, 1997. **8**(1): p. 23-42.
- 31. Massetti, B. and R.W. Zmud, *Measuring the extent of EDI usage in complex organizations: Strategies and illustrative examples.* MIS Quarterly: Management Information Systems, 1996. **20**(3): p. 331-345.
- 32. Zhu, K., et al., *Migration to open-standard interorganizational systems: Network effects, switching costs, and path dependency.* MIS Quarterly: Management Information Systems, 2006. **30**(SPEC. ISS.): p. 515-539.

- 33. Schwartz, R.B. and M.C. Russo, *How to Quickly Find Articles in the Top IS Journals*. Communications of the ACM, 2004. **47**(2): p. 98-101.
- 34. Davila, A., M. Gupta, and R.J. Palmer, *Moving procurement systems to the internet: The adoption and use of e-procurement technology models.* European Management Journal, 2003. **21**(1): p. 11-23.
- 35. Starrenburg, M. and J.H.A. Harink. *Supplier Relationship Management*. IPSERA Papers 2006 [cited 1-9-2007]; Available from: http://www.ht2.org/conference/pdf/155.pdf.
- 36. Kim, J.-I. and D.L. Shunk, *Matching indirect procurement process with different B2B e-procurement systems*. Computers in Industry, 2004. **53**(2): p. 153-164.
- 37. Wang, S. and N. Archer, *Business-to-business collaboration through electronic* marketplaces: An exploratory study. Journal of Purchasing and Supply Management, 2007. **13**(5-6): p. 113-126.
- 38. Chatterjee, D. and T. Ravichandran, *Beyond exchange models: Understanding the structure of B2B information systems*. Information Systems and e-Business Management, 2004. **2**: p. 169-186.
- 39. Ravichandran, T., S. Pant, and D. Chatterjee, *Impact of industry structure and product characteristics on the structure of B2B vertical hubs*. IEEE Transactions on Engineering Management, 2007. **54**(3): p. 506-521.
- 40. Chatterjee, D. and T. Ravichandran. Inter-organizational information systems research: A critical review and an integrative framework. in Proceedings of the Hawaii International Conference on System Sciences. 2004.
- 41. Bui, T., A. Gachet, and H.J. Sebastian, *Web services for negotiation and bargaining in electronic markets: Design requirements, proof-of-concepts, and potential applications to e-procurement.* Group Decision and Negotiation, 2006. **15**(5): p. 469-490.
- 42. Kumar, K. and H.G. Van Dissel, *Sustainable collaboration: Managing conflict and cooperation in interorganizational systems*. MIS Quarterly: Management Information Systems, 1996. **20**(3): p. 279-299.
- 43. Goldkuhl, G., *The six Phases of Business Processes Business Communication and the Exchange of Value*, in *The 12th Biennal ITS conference ITS'98*. 1998: Stockholm.
- 44. Robinson, P.J., C.W. Faris, and Y. Wind, *Industrial Buying and Creative Marketing*. 1967, Boston: Allyn & Bacon.
- Boyson, S., T. Corsi, and A. Verbraeck, *The e-supply chain portal: A core business model*. Transportation Research Part E: Logistics and Transportation Review, 2003. 39(2): p. 175-192.
- 46. Barratt, M. and A. Oliveira, *Exploring the experiences of collaborative planning initiatives*. International Journal of Physical Distribution & Logistics, 2001. **31**(4).
- 47. Supply-Chain Operations Reference-Model version 8.0. 2006, Supply-Chain Council.

- 48. Kilger, C. and B. Reuter, *Collaborative Planning*, in *Supply Chain Management and Advanced Planning*, H. Stadtler and C. Kilger, Editors. 2005, Springer: Berlin.
- 49. Markus, M.L. and D. Robey, *Information Technology and Organizational Change: Causal Structure in Theory and Research*. Management Science, 1988. **34**(5): p. 583-598.
- 50. Tornatzky, L.G. and M. Fleischer, *The Process of Technology Innovation*. 1990, Lexington, MA: Lexington Books.
- 51. Chwelos, P., I. Benbasat, and A.S. Dexter, *Research Report: Empirical Test of an EDI Adoption Model.* Information Systems Research, 2001. **12**(3): p. 304-321.
- 52. Singh, N., K.H. Lai, and T.C.E. Cheng, *Intra-organizational perspectives on ITenabled supply chains*. Communications of the ACM, 2007. **50**(1): p. 59-65.
- Henderson, J.C. and N. Venkatraman, Strategic alignment: leveraging information technology for transforming organizations. IBM Systems Journal, 1993. 32(1): p. 4-16.
- 54. Son, J.Y., S. Narasimhan, and F.J. Riggins, *Effects of relational factors and channel climate on EDI usage in the customer-supplier relationship.* Journal of Management Information Systems, 2005. **22**(1): p. 321-353.
- 55. Tanner, C., et al., Current Trends and Challenges in Electronic Procurement: An Empirical Study, in 20th Bled eConference eMergence: Merging and Emerging Technologies, Processes, and Institutions. 2007: Bled, Slovenia.
- 56. Iacovou, C.L., l. Benbasat, and A.S. Dexter, *Electronic Data Interchange and Small Organizations: Adoption and Impact of Technology*. MIS Quarterly, 1995. **19**(4): p. 465-485.